1.0 ADMINISTRATIVE BYLAWS

1.1 GENERAL PROVISIONS

- 1.1.1 The following Bylaws shall regulate the administration of the affairs of the Town of Bolton, Massachusetts.
- 1.1.2 Any of the Bylaws under 1.0 may be repealed or amended or other Bylaws may be adopted by a majority vote, at any town meeting, an article or articles for that purpose having been inserted in the warrant for such meeting.
- 1.1.3 The Selectmen shall publish the Bylaws of the town at least once every five years.
- 1.1.4 Whoever violates any of the provisions of these Bylaws whereby any act or thing is enjoined or prohibited, shall, unless other provision is expressly made, forfeit and pay a fine of not more than Twenty Dollars (\$20.00) for each offense. Each day of violation shall constitute a separate offense.
- 1.1.5 The preceding sections notwithstanding, any person violating the provisions of any section of these bylaws which is subject to a specific penalty may be penalized by a non-criminal disposition as provided in M.G.L. Chapter 40 s 21D which includes the giving, to the violator, a written notice to appear before the clerk of the district court. The non-criminal method of disposition may also be used for violations of any rules or regulations of any town officer, board or department which is subject to a specific penalty. Without intending to limit the foregoing, the following bylaws, rules and regulations are to be included within the scope of this section, the specific penalties listed shall apply and in addition to police officers, who in all cases shall be enforcing officers, the Town personnel listed shall also be enforcing officers:

a) General Protective and Regulating Provisions of Section 1.9

Penalty: \$20.00 a day

Enforcing Officer:

Section 1.9.1	(nuisance dogs and kennels) – Animal Control Officer
Section 1.9.2	(hawkers and peddlers) - Police Department
Section 1.9.3	(junk dealers) - Police Department
Section 1.9.4	(livestock at large) - Animal Control Officer
Section 1.9.5	(tethering to a tree) - Director of Public Works
Section 1.9.7	(lumber piles) - Director of Public Works
Section 1.9.8	(digging in highway) - Director of Public Works
Section 1.9.10	(unregistered motor vehicle) - Police Department
Section 1.9.13	(town beach) - Police Department
Section 1.9.14	(motorized recreation vehicle) - Police Department
Section 1.9.19	(hunting and trapping) – Police Department

- b) Removal of Soil, Loam, Sand and Gravel Bylaw (Section 1.10) Penalty: \$50.00 a day Enforcing Officer: Board of Selectmen
- c) Protection of Groundwater Bylaw (Section 1.11) Penalty: \$50.00 a day Enforcing Officer: Board of Health
- d) Street Numbering Bylaw (Section 1.13) Penalty: \$10.00 a day Enforcing Officer: Police Department
- e) Violation of any provision of the Zoning Bylaws Penalty: \$50.00 a day Enforcing Officer: Board of Selectmen
- f) Violation of any condition of a Special Permit of Variance granted pursuant to the Zoning Bylaw or M.G.L. Chapter 40A. Penalty: \$50.00 a day Enforcing Officer: Board of Selectmen
- g) Discharge of Snow or ice onto Public Ways (Section 1.19)
 Penalty: \$100
 Enforcing Officer: Police Department
- h) Violation of Wetland Bylaw (Section 1.18)
 Penalty: \$300.00 for each offense, each day of violation shall be a separate offense. Enforcing Officers: Conservation Commission and Agent

1.2 TOWN MEETINGS AND WARRANTS

- 1.2.1 The Annual Town Meeting for the transaction of business shall be on the first Monday in May at a time to be determined by the Board of Selectmen. The election of officers shall take place on the second Monday in May and commence at 12:01 p.m. and close at 8:00 p.m. of that day.
- 1.2.2 The Annual Town Meeting shall, unless a different time or method is prescribed by law, be called by posting an attested copy of the warrant, calling the same, at the Town Hall, at the Post Office and at two other public places in the Town seven days, at least, before the day appointed for said meeting, and by mailing a copy of the warrant to each household where one or more registered voters resides.

A Special Town Meeting shall, unless a different time or method is prescribed by law, be called by posting an attested copy of the warrant, calling the same, at the Town Hall, at the Post Office and at two other public places in the town fourteen days, at least, before the day appointed for said meeting, and by mailing a copy of the warrant to each household where one or more registered voters resides.

- 1.2.3 The presence of seventy-five (75) registered voters at a town meeting for the transaction of business shall be required to constitute a quorum except for a motion to adjourn for which no quorum shall be required.
- 1.2.4 When an article requiring an appropriation of money is presented in the warrant for a town meeting, said article must contain the request for the appropriation.

For all matters requiring a two thirds vote under Chapter 39, Section 15 of the Mass General Laws, a count need not be taken unless the vote so declared is immediately questioned by seven or more voters.

- 1.2.5 The Moderator shall have the power to require persons who are not registered voters to identify themselves and leave the floor of the hall when he shall deem this action necessary.
- 1.2.6 Articles for the warrant shall be acted upon in the order in which they stand, except that the Moderator may upon request and for reasons stated entertain the motion to take up an article out of this regular order.
- 1.2.7 No motion, the effect of which would be to dissolve a town meeting, shall be in order until every article in the warrant has been acted upon, but this shall not preclude the postponement of action on, or consideration of, any article to an adjournment of the meeting to a stated time.

When a question is before the meeting, the following motions, namely: to adjourn; to lay on the table; for the previous question; to postpone to a stated time; to commit, recommit or refer; to amend; to postpone indefinitely; shall be received and shall have precedence in the foregoing orders, and the first three shall be decided without debate.

No article in the warrant shall be again taken into consideration after having been disposed of unless ordered by a vote of two-thirds of the voters present and voting.

- 1.2.8 The Moderator may decline to put motions obviously frivolous or tending to disorder. A motion shall be presented in writing if the Moderator requests. The Moderator shall be governed in his rulings by these articles.
- 1.2.9 All committees unless otherwise specially directed by the meeting shall be appointed by the Moderator except as otherwise provided by law and all committees so appointed shall be directed to report within a definite time. If a committee does not report within the time stated, or at the first Annual Town Meeting thereafter, it shall be considered discharged. The Moderator shall not be a member of any committee appointed by him.

1.2.10 It shall be the duty of the appointing body promptly after every town meeting to notify all members of committees who shall be selected or appointed at such meeting.

1.3 ADVISORY COMMITTEE

- 1.3.1 The Advisory Committee shall consist of six citizens of the Town, which committee shall be appointed as provided in the following section, and no person holding an elective or appointive town office shall be eligible to serve on said committee.
- 1.3.2 The Moderator, Town Clerk and Chairman of the Board of Selectmen shall constitute the appointing committee, who shall, within 30 days of each annual meeting for the election of officers appoint from the citizens of the Town two members to serve for the term of three years. The terms of office of said members shall expire June 30. Said committee shall choose its own officers, and shall serve without pay excepting, however, the secretary thereof, who may receive such compensation as the Town may by vote provide. Said committee shall cause to be kept a true record of its proceedings.
- 1.3.3 To the Advisory Committee shall be referred for recommendations all articles in any warrant for a town meeting hereafter issued. It shall be the duty of the Board of Selectmen after drawing any warrant for a town meeting, to transmit immediately a copy thereof to the Chairman of the Advisory Committee, and said committee shall consider such articles. A public hearing on said articles may be held at the discretion of said committee, unless a public hearing by some other tribunal is required by law, and a notice of such hearing shall be given by posting a copy thereof in at least three public places in town. After due consideration of the subject matter in such articles, said committee shall report thereon, in print or otherwise, such information and recommendations as it shall deem best. All recommendations of the committee made to the Town shall be recommendations of a majority of the committee. In case of disagreement, all reports shall be reported back to the Town without recommendation. To the Advisory Committee shall also be referred the disposition of all tangible supplies and property, not including real property, the value of which exceeds \$1,000. The Advisory Committee shall dispose of all such supplies and property in compliance with applicable procurement requirements. The agency possessing tangible supplies and property, not including real property, the value of which is \$1,000 or less shall dispose of such supplies and property using sound business principals.
- 1.3.4 It shall be the duty of the Advisory Committee to consider the annual estimates and expenditures as prepared by the various town officers, boards, and committees charged with the expenditure of the Town's money, and add another column to the prepared statement giving the amount which in its opinion shall be appropriated for the ensuing year, and add thereto such explanations and suggestions in relation to the proposed appropriations as it may deem expedient, and report thereon, as provided in 1.3.3.

The Selectmen shall include in their estimates, separately, the salaries of all officers elected directly by the Town, and all other items not otherwise provided for, and the salaries of all other persons shall be included in the estimates of officers, boards, or committees appointing them. The Treasurer shall include in his estimates the amount required for the payment of interest on the town debt and for the payment of such portions of the town debt as may become due during the succeeding year. Said estimates and explanations shall be filed with the Advisory Committee on or before the fifteenth day of January.

- 1.3.5 Whenever any vacancy shall occur in said committee by resignation, removal from town, death, failing to qualify, or otherwise said vacancy shall be filled by said committee. And if any member is absent from five consecutive meetings of said committee, except in case of illness, said committee shall consider the position vacant and proceed to fill the same. The term of office of all persons chosen as aforesaid, to fill vacancies, shall expire on June 30, succeeding such vacancy and a successor shall be appointed to fill out the unexpired term of each member whose office has been so vacated, in the same manner as the original appointment.
- 1.3.6 It shall be the duty of said committee to make an annual report of its doings, with recommendations relative to financial matters, to be printed with the annual reports of the other town officers.
- 1.4 LEGAL AFFAIRS
- 1.4.1 The Selectmen shall have the general direction and management of the property and affairs of the Town in all matters not otherwise provided for by law or by these Bylaws.
- 1.4.2 The Selectmen together with two other citizens of the Town neither of whom shall hold an elective town office, shall constitute a Law Committee. The two members holding no elective office so serving with the Selectmen shall be appointed annually by the Moderator. This committee, except as otherwise provided by law or vote of the Town, shall have full and exclusive authority as agents of the Town to engage counsel, to institute, prosecute, defend, compromise and settle all claims, suits and actions brought by or against the Town or protect the Town with liability insurance from any or all claims, provided, however, that no claim or action against the Town, unless reduced to the form of an execution or decree of Court shall, except in cases which are covered by liability insurance, be compromised or settled by the payment of any amount in excess of Two Hundred Dollars (\$200.00) without a special vote of the Town.
- 1.4.3 It shall be the duty of the Law Committee to notify the citizens of the Town by posting in four public places as soon as possible notice of the fact that an action at law or suit in equity has been brought against the inhabitants of the Town,

when the amount of the action or suit is Five Hundred Dollars (\$500.00) or more.

- 1.4.4 All conveyances of land or interests in land which may hereafter be authorized by vote of the Town, or otherwise, except land held under tax titles, shall be signed by a majority of the Board of Selectmen unless otherwise provided by law, or these Bylaws or by special vote of the Town, and the same shall be sealed with the town seal.
- 1.4.5 The Law Committee in their annual report shall state what actions have been brought against or on behalf of the Town, what cases have compromised or settled and the terms thereof, and the current standing of all suits of law involving the Town or any of its interests, and they shall give a summary of their activities and decisions during the past year.

1.5 FINANCIAL AFFAIRS

1.5.1 The Town Treasurer shall prescribe the methods of accounting and the forms to be used by the several officers, boards and committees of the Town pertaining to their receipts and disbursements, and shall provide that such methods and forms shall conform to the requirements prescribed by law or by any rules or regulations made thereon.

1.6 PURCHASING AND CONTRACTS

1.6.1 All procurement, for which appropriations have been made or monies otherwise legally provided, shall be made in compliance with MGL, c. 30B, the Uniform Procurement Act; MGL c. 7, ss. 38A ¹/₂-38O, Design Services; MGL c. 149, ss. 44A-44J, Public Building Construction; MGL c. 30, ss. 39A-40A, Public Works Construction; and all other relevant laws, as they may be amended from time to time. It shall be the responsibility of each department and designated committee to determine if funds are available before making commitments.

Each expenditure shall be approved by the head of the department or a majority of the committee having control. Bills presented by vendors for services shall be approved and signed by a majority of the department officers or committee responsible and also by a majority of the Board of Selectmen. Bills shall then be passed to the Town Treasurer for payment.

It shall be the responsibility of each department or committee to keep total expenditures within monies legally available. The Town Treasurer shall pay only up to the total amount authorized for the designated purpose or department or committee requirements.

- 1.6.2 Deleted April 9, 2001, Article 2
- 1.6.3 Deleted April 9, 2001, Article 2

1.6.4 Any person or firm requiring the services of the Tree Warden and/or Moth Superintendent shall pay for services at an hourly rate equal to the hourly rate of the General Foreman of the Forestry Division Department of Public Works. Said compensation shall be paid directly to said Tree Warden and/or Moth Superintendent.

1.7 RECORDS AND REPORTS

- 1.7.1 All officers, boards and committees of the Town shall cause records of their doings and accounts to be kept in suitable books. Said books shall be kept in their respective places in the Town, and shall not be removed therefrom. Said books, shall unless otherwise provided by law, be open to public inspection at any reasonable time, but shall remain during such inspection under supervision of the officer, board or committee having custody thereof.
- 1.7.1.1 All boards, committees, commissions of the Town, elected or appointed, shall compile and maintain minutes of all posted and emergency meetings and shall file a copy of said minutes with the Town Clerk within one week of approval. Minutes of executive sessions need not be filed with the Town Clerk until such time as the reason for the session is no longer valid.
- 1.7.2 All officers, boards and standing committees, and special committees of the Town having charge of the expenditure of town money shall annually report thereon in writing so as to give the citizens a fair and full understanding of the objects and methods of such expenditures, referring, however, to the report of the Town Treasurer for statements in detail of receipts and payments, and may make therein such recommendations as they deem proper. Such reports shall be submitted to the Selectmen for inclusion in the Annual Town Report on or before the l0th day of January of each year. All reports shall be signed by at least a majority of the board. The Selectmen shall cause these reports, together with a list of town officers and other matter usually published, to be printed and placed in the hands of the Town Clerk, ready for distribution, two months, at least, before Annual Town Meeting in May.
- 1.7.3 The Annual Town Report shall contain, in addition to the reports of officers, boards and committees as herein before provided, a detailed report of all monies received into and paid out of the town treasury in the year next preceding, showing separately payments made from the proceeds of loans as capital outlays for permanent improvements, the report of the Collector of Taxes, of receipts, payments and statements; statements of all funds belonging to the Town or held for the benefit of its inhabitants; a statement of the liabilities of the Town on bonds, notes, certificates of indebtedness authorized but not incurred, and the purposes thereof; a statement of transfers made to or from any appropriation; abstracts of the records of the meetings of the Town held since the publication of the last annual report; and such other matters as the said report is required by law.

- 1.7.4 The several boards of town officers shall file with the Board of Selectmen at the end of each calendar year a schedule of all property in their charge. These schedules shall be open to public inspection.
- 1.7.5 The Selectmen, or the Town may direct that the Assessors' valuation list, the Bylaws and standing votes of the town and the rules and regulations adopted by any officer, board or committee, be printed either separately or as a part of the annual report.
- 1.7.6 The Town Clerk shall conduct an annual census.

1.8 CEMETERIES

- 1.8.1 The Town of Bolton will accept from any person any sum of money not less than One Hundred Dollars (\$100.00), the interest whereof shall be applied to the care and repair of such lot or lots in the public burial ground of the Town as the person depositing the money shall designate, in accordance with Section 25 of Chapter 114 of the General Laws (Ter. Ed.).
- 1.8.2 All such money shall be deposited by the Town Treasurer to the credit of the burial fund. The Treasurer shall keep the account of burial lot fund in a book showing the date of each deposit, the amount deposited, the name of the depositor, the name of the cemetery containing the lot and the number of the lot.

1.8.3 The treasurer shall give the depositor a receipt in the following form:

	Date
This certifies	that has deposited with me the sum of
	Dollars, the interest of which is to be forever applied in
accordance wit	th the provisions of Section 25, Chapter 114, General Laws (Ter.
	care of lot number on the plan of the
	Cemetery.
	Treasurer, Town of Bolton

- 1.8.4 The Cemetery Committee shall not be required to expend the interest on each deposit annually but may, at its discretion, allow it to accumulate in order to do more satisfactory work.
- 1.8.5 Lots in the cemeteries shall be free to inhabitants of the Town of Bolton.
- 1.8.6 The assignment of lots shall convey only the right of burial. The Title to all lots shall be vested in the Town.
- 1.8.7 No lots shall be graded, nor shrubbery or trees planted, nor monuments erected, in any cemetery, except with the approval of the Cemetery Committee.

1.8.8 Any grave opened in Bolton cemeteries shall be paid for at the rate of no less than \$150.00 for regular hours. The rate for overtime or weekends shall be no less than \$250.00 and holiday hours shall be no less than \$500.00.

Winter burial shall be at the discretion of the Cemetery Committee and if done with extra equipment (i.e. compressor, plowing, sanding) shall be paid for by the deceased's account.

This money to revert to the Town Treasury and shall be used in accordance with Sections 15 and 25 of Chapter 114 of the General Laws (Ter.Ed.).

- 1.8.9 Within thirty (30) days after the closing of a grave a flush marker shall be placed on such grave under the direction of the Cemetery Committee.
- 1.9 PROTECTIVE AND REGULATING PROVISIONS
- 1.9.1 DOG AND KENNEL BYLAW
- 1.9.1.1 The Town of Bolton adopts this Dog and Kennel Bylaw in accordance with, and to conform with, Sections 136A through 174D of Chapter 140 of the Massachusetts General Laws.
- 1.9.1.2 For the purposes of this Section 1.9.1 "kennel" shall be defined as one (1) pack or collection of more than three (3) dogs six (6) months old or older on a single premises, whether maintained for breeding, boarding, sale, training, hunting, or other purposes.
- 1.9.1.3 At no time shall a person owning or keeping a dog in the Town allow such dog to be unlicensed, uncollared, untagged, or to become a "public nuisance" to others within the Town Limits. A dog shall be deemed a "public nuisance" for, but not limited to, the follow reasons:
 - attacking or chasing persons or domestic animals
 - damaging property of others beyond its owner/keepers property
 - interfering with or disrupting organized school or public activities
 - interfering with the use of a public right-of-way
 - chasing moving vehicles
 - vicious disposition
 - barking or howling excessively
 - creating other disturbance
- 1.9.1.4 Every person owning or keeping a dog that is six (6) months old or older shall obtain a dog license from the Town Clerk, and every person maintaining a kennel shall obtain a kennel license from the Town Clerk, as required by Section 137 and 137A of Chapter 140 of the Massachusetts General Laws. Every such dog and kennel license shall be valid for one year beginning on April first and ending on March thirty-first. Each license shall contain the name, address, and telephone number of the owner or keeper of the dog, and such other information as the Town Clerk deems appropriate; in addition, each kennel

license shall include the number of dogs maintained in the kennel. Each license shall be issued upon the condition that the owner or keeper shall comply with the provisions of this bylaw and any law, rule, or regulation relating to the ownership and control of dogs.

- 1.9.1.5 As allowed by Section 173 of Chapter 140 of the Massachusetts General Laws, all owners or keepers of dogs kept in the Town of Bolton, who on the first day of May of each year have not licensed said dog or dogs as prescribed by Section 137 and 137A of Chapter 140 of the Massachusetts General Laws, and this bylaw, shall be required to pay an additional fee of twenty-five (\$25.00), payable to the Town of Bolton. If still unlicensed, uncollared, and untagged after June 1st the Board of Selectmen shall, by June 11th, issue a warrant directing the Animal Control Officer to seek out, catch, and confine all such dogs; and to enter and prosecute a complaint in district court against the owners or keepers, if known, of said dogs, in accordance with said Chapter 140, Section 151A.
- 1.9.1.6 Every holder of a kennel license shall maintain the kennel in a sanitary and humane manner, and shall keep a record of all dogs kept in the kennel, including their license numbers.
- 1.9.1.7 The Board of Selectmen, the Chief of Police, or the Animal Control Officer, or their designees, may at any time inspect, or cause to be inspected, any kennel licensed under these kennel provisions, and if, in their judgment, the kennel is not being maintained in a sanitary or humane manner, or if records are not properly kept as required by this bylaw or other applicable law, or if the kennel is in violation of any provision of this Section 1.9.1, the Board of Selectmen shall by order revoke or suspend, and, in the case of suspension, may reinstate, such license.
- 1.9.1.8 Notwithstanding the provisions of Section 147 of Chapter 140 of Massachusetts General Laws or any other provision of the law to the contrary, all money received from the sale of licenses for dogs, or recovered as fines or penalties by the Town under the provisions of Chapter 140 of the Massachusetts General Laws relating to dogs, shall be deposited in the Town treasury in a special fund to be known as the Dog Fund. The money in the Dog Fund shall be expended at the direction of the Town Clerk and/or Town Treasurer without further appropriation for the payment of costs incurred in administering the dog licensing program, and for the payment of costs and expenses imposed on the Town by (a) Section 151B of Chapter 140 of the Massachusetts General Laws for emergency care, treatment, or disposal by registered veterinarians of dogs and cats injured on ways; and (b) Section 161 of Chapter 140 of the Massachusetts General Laws for damage to livestock and fowl by dogs, and for appraising the amount of said damage, provided that such damage shall not exceed the amount of one thousand dollars (\$1,000) for each occurrence. Upon the location of the owner(s) of the offending animal(s), said owner(s) shall reimburse the Dog Fund for said expenses.

- 1.9.1.9 Violations of this section 1.9.1 or Massachusetts General Laws Chapter 140, Section 136A through 174D shall be subject to impoundment of the animal by the Animal Control Officer, and/or a fine as referenced in Bolton Bylaws Section 1.1.5; and/or by criminal or non-criminal disposition as provided in said Chapter 140, Sections 151A, 157, and 173A. Any violation resulting in impoundment shall also require payment to the Town of Bolton of a pick-up fee of ten dollars (\$10), plus normal boarding fees.
- 1.9.2 No person shall hawk or peddle fish, fruit or vegetables within the limits of the Town, except as otherwise authorized by law, without obtaining a license therefore from the Selectmen upon payment of a fee at a rate to be set annually by the Selectmen.
- 1.9.3 No dealer in junk, old metals or second hand articles, shall collect or keep such articles in the Town of Bolton except as otherwise authorized by law, without a license from the Board of Selectmen. The Selectmen, after notice, and hearing, may revoke any such license for cause. The fee for such license shall be at a rate set annually by the Selectmen.
- 1.9.4 No livestock, including but not limited to horses, swine, sheep, goats, cattle, or llamas shall be allowed to run at large in any roadway of the Town.
- 1.9.5 No person shall tie a horse or other animal to any tree, nor to any structure protecting such tree, in the public streets of the Town.
- 1.9.6 (Moved to 2.4.2 on 6/12/78)
- 1.9.7 No person shall place or pile any wood, lumber or other material within the limits of the highway or sidewalks without a written permit from the Selectmen, and then only for a limited period.
- 1.9.8 No person except officers of the Town and their duly authorized agents in the lawful performance of their duties, shall break up or dig up the ground in any highway of the Town, or set up any post, pole, fence, tree, structure, or any other obstruction, without a written permit from the Selectmen.
- 1.9.9 Any project involving major construction and/or taking of public or private property shall be explained to the inhabitants of the town through a Public Hearing. Such Public Hearing shall be called by the Selectmen by posting notices thereof in three public places including the outside of the Town Hall at least seven days before the time of said hearing.
- 1.9.10 Unregistered motor vehicles which are unfit for use, permanently disabled, or have been dismantled or are otherwise inoperative, shall not be stored, parked, or placed upon any land in the Town unless the same shall be within a building or in an area unexposed to the view of the public or abutters or are in an area properly approved for the keeping of the same by licensed junk dealers or automobile dealers.

- 1.9.11 The penalty for violation of the provisions of this Article shall not be less than Ten Dollars (\$10.00) nor more than Twenty Dollars (\$20.00) for each offense. Each day that each violation continues shall constitute a separate offense.
- 1.9.12 (Moved to 2.5.2.2 on 6/12/78)
- 1.9.13 The Selectmen shall from time to time establish regulations as necessary to insure safe and proper operation and conduct at the Town Beach, such regulations to be enforced by the Police Department and failure to comply shall be subject to fines under this Division.
- 1.9.14 No person shall use or operate a motorized recreation vehicle including motorcycles, trail bikes, minibikes, snow vehicles, all-terrain vehicles, and similar types, on public or private property, within the confines of the Town of Bolton, except with the consent of the owner thereof or his duly authorized representative. In the case of public land, consent shall be obtained from the appropriate Town Board.
- 1.9.15 The practice of going in and upon private residences of the town by commercial agents, selling agents, solicitors and canvassers, transient vendors and itinerant merchants for the purpose of soliciting orders for services or for the sale of goods, wares, and merchandise by means of samples, lists, catalogues or otherwise, without having been requested or invited to do so by the owner or occupant of said private residences, is prohibited and hereby declared a nuisance.

The provisions of this Bylaw shall not apply to officers or employees of the Town, County, State or Federal governments; hawkers and peddlers registered by the State and the Town under appropriate laws and regulations, candidates for public office or political parties recognized by the Commonwealth; religious organizations for the purpose of spreading the teachings of their religious beliefs, but not for the purpose of selling or soliciting; and nonprofit, charitable organizations upon registration by the President or Treasurer with the Chief of Police. Such registrations must be renewed yearly.

1.9.16 Gas Inspector

The Board of Selectmen shall appoint each year an Inspector of Gas Piping and Gas Appliances whose duty shall be the enforcement of the rules and regulations adopted by the board established under Section 12H of Chapter 25 of the General Laws.

1.9.17 Plumbing Inspector

The Inspector of Buildings in accordance with the provisions of Chapter 142, Section 11 of the General Laws, shall appoint annually an Inspector of Plumbing, who shall carry out his duties as set forth in said statute.

- 1.9.18 Inspector of Buildings The Board of Selectmen shall appoint each year an Inspector of Buildings. His responsibilities are described in Section 2.1.2 Administration.
- 1.9.19 Hunting, Trapping and Firearms Regulations
 - a) No person shall fire or discharge any firearms, or set traps, on any property without the written consent of the owner or legal occupant.
 - b) No person shall fire or discharge any firearm, hunt, or set traps, on any public property without the written consent of the officials, or their agent for this purpose, charged with the management of the particular public property. For the purposes of this bylaw, regulations promulgated by state agencies having jurisdiction over public property within the Town of Bolton specifically authorizing hunting, fishing thereon shall constitute written consent.
 - c) This Bylaw shall not apply to the lawful defense of life and property nor any Law Enforcement Officer in the performance of his duties.
- 1.9.20 Fire Protection Systems Pursuant to Section 98 of Chapter 143 of the General Laws, the Selectmen may adopt regulations relative to fire protection systems with the approval of the Fire Chief after consultation with the Planning Board.
- 1.9.21 The Town of Bolton may use employees of the Highway Department as flag persons for traffic control during any road maintenance or repair project being performed by the Town.
- 1.10 REMOVAL OF SOIL, LOAM, SAND AND GRAVEL
- 1.10.1 The removal of soil, loam, sand or gravel from any parcel of land not in public use in the Town of Bolton, except as hereinafter provided, shall be allowed only after a written permit therefore is obtained from the Board of Selectmen after a Public Hearing of which due notice is given.
- 1.10.2.1 The Board of Selectmen shall fix a reasonable time for the hearing for a removal permit and shall cause the notice of the time and place of such hearing thereof and of the subject matter, sufficient for identification, to be published in a newspaper of general circulation in the Town. Such publication to be not less than seven (7) days before the day of the hearing, or by posting such notice in a conspicuous place in the Town for a period of not less than seven days before the day of such hearing, and also send notice by mail, postage prepaid, to the petitioner and to the owners of all property deemed by the Board to be affected

thereby including the abutters notwithstanding that the abutting land is located in another city or town, as they appear on the most recent tax lists. The publication required by this section shall contain the following printed in bold face type:

- a) the name of the petitioners;
- b) the location of the area or premises which are subject of the petition; and
- c) the date and place of the Public Hearing.
- 1.10.2.2 No permit shall be required for the removal of soil, loam, sand or gravel from any parcel of land when incidental to and in connection with the construction of a building on the parcel. No permit shall be required by a home owner or farmer for the transfer of loam, sand, or gravel from one part of his own lot to another part thereof for the improvement of his own lot. Nor shall any permit be required for removal of loam, sand or gravel in connection with the improvement or construction of any road over a lot provided such loam, sand or gravel so removed is used on said lot to improve the same and is not contrary to the regulations of this Bylaw as adopted. No permit shall be required by quarries operating at the time this Bylaw is adopted so long as the sale of soil, loam and gravel is not involved.
- 1.10.2.3 A permit shall be issued without hearing by the Board of Selectmen for the removal of sand or gravel from any lot within the Town wherein the removal operation can be proven by the owner to said Board to have been in continuous operation since October 10, 1956.
- 1.10.2.4 All permits in force as of the effective date of this Bylaw, are subject to said Bylaw and the holders thereof are required to submit applications for permits in accordance with the terms of this Bylaw for hearings to be held at least one (1) week prior to the anniversary date of the presently issued permit. Permits will be granted as recited in this Bylaw for a period not to exceed three years and all procedures shall be followed as set forth in this Bylaw concerning hearings prior to the issuance of permits.
- 1.10.3 In issuing a permit under this Bylaw, the Board of Selectmen may impose such conditions not specifically provided therein as it may deem necessary for the adequate protection of the neighborhood and the Town. Any conditions imposed by the Board shall be attached to and made a part of the permit. The Board may, in its discretion, require a bond, certified check, or other security for compliance with such conditions or as evidence of good faith as to the completion of any proposed construction. The Board may, after a Public Hearing on proof of violation of any condition, revoke any permits so issued. No permit shall be issued under the provisions of this Bylaw for a period of more than three years.
- 1.10.4 Sand and gravel may be removed from any parcel of land except within 300 feet of a street or way, and the Board shall issue a permit therefore provided, however, that the Board shall impose such reasonable conditions as to the

disposition of topsoil and the re-establishment of ground levels and grades as it may deem necessary.

- 1.10.5 Soil or loam may be removed and sold from any parcel of land within such parcel determined by the Board to be unsuited to agricultural use, and the Board may issue a permit for such removal, provided, however; that the Board shall in making such decision, obtain the recommendations of the appropriate Soil District Supervisor and the County Extension Director or Agent or their successors, and their recommendations shall be made a part of the records of the Board. In issuing a permit, the Board may impose reasonable conditions as to the re-establishment of ground levels and grades.
- 1.10.6 Notwithstanding the provisions of the above, the Board may issue a permit for the removal of soil or loam from any parcel of land in the Town where such removal is necessarily incidental to and in connection with the construction of a road or other facility involving a permanent change in the use of the land. The Board shall issue no such permit unless it is reasonably satisfied that the construction will be completed and evidence thereof shall be made part of the records of the Board.
- 1.10.7 Sand or gravel may be removed from any parcel of land within such parcel lying within 300 feet of any street or way, provided a permit therefore has been issued by the Board after satisfactory evidence that such removal will not be seriously detrimental or injurious to the neighborhood provided further that the Board shall impose reasonable conditions as to the method of removal, the re-establishment of ground levels and grades and the planting of the area to suitable cover, as it may deem necessary.
- 1.10.8 The penalty for violation of the provisions within this division (or article) shall be as follows. For each offense: Fifty Dollars (\$50.00); and each day that each violation continues shall constitute a separate offense.

1.11 **PROTECTION OF GROUNDWATER**

In order to protect, preserve, and maintain the existing and potential groundwater supply and groundwater recharge areas within the Town of Bolton, no person shall alter the quality of any surface water or groundwater without first filing with the Board of Health and obtaining and complying with a groundwater alteration permit. For the purposes of this Bylaw, the following activities are allowed without permit:

- a) Application of fertilizer and pesticides in accordance with manufacturers' recommended procedures or standards approved by state, federal, or local agency.
- b) Non-injection sewage disposal (e.g. typical domestic septic systems) in accordance with Title 5 of the State Environmental Code.
- c) Construction of single family dwellings, additions and accessory buildings.

This Bylaw should in no way be interpreted to limit or restrict the authority of the Board of Health.

1.11.1 Permit

The Board of Health shall issue and improve regulations which shall govern filing procedures and other matters related to carrying out the requirements of this Bylaw.

The Board of Health shall, within 21 days of receipt of all required information, determine at a public meeting whether the proposed alteration will have a significant effect on the groundwater quality, groundwater recharge or groundwater elevation. If the Board of Health determines that insufficient information has been provided by the applicant or that additional technical consultation is required it may delay the determination until the required information and/or consultation can be provided.

1.11.2 Performance Standards

Any alteration subject to the permit requirements of this Bylaw shall comply with the following minimum performance standards.

- a) All hazardous material shall be retained in product-tight containers and removal and disposal off-site shall be directed by the Board of Health. No hazardous material shall be present in wastes disposed of on-site.
- b) Process wastes from operations other than personal hygiene and food for residents, patrons, and employees shall be treated so that contaminant levels in groundwater resulting from such disposal will not exceed background levels of individual constituents. Process wastes shall be disposed of in a separate location from wastes generated from personal hygiene and food for residents, patrons, and employees or from building and parking area drainage.
- c) All chemicals, fuel, fertilizers, pesticides or other potentially contaminating substances shall be stored and handled in such a way that will prevent the release of leachate to (surface or) groundwater. Storage and handling measures shall include protection from vandalism, accidental damage or corrosion.
- d) All underground tanks shall meet the initial and periodic inspection requirements as promulgated in Board of Health underground tank regulations. Underground tanks of 300 gallons or less shall not be inspected more than once every four years.
- e) All runoff from artificial impervious surfaces shall be recharged where possible on the site, diverted toward areas covered with vegetation for surface infiltration. Dry wells shall be used only where the above methods are infeasible, and shall be preceded by grit chambers to facilitate removal of contaminated solids. No discharge directly into surface waters without

intervening mitigative measures, including trapped catch basins and oil absorbents, will be allowed.

- f) New commercial earth removal operations shall be limited to a minimum of 4 feet above maximum ground water and exposed land shall be returned to its natural vegetative state as of the date of implementation of the Bylaw when excavation is complete.
- g) An erosion and sedimentation plan shall be provided to control the effects of construction to maximize on-site recharge.
- 1.11.3 The Board of Health, its agents, offices and employees may enter upon the land on which an alternative is proposed in response to a filing or application, and/or for the purposes of carrying out its duties under this Bylaw, and may make or cause to be made such examination or survey as it deems necessary.
- 1.11.4 This Bylaw shall not apply to emergency projects necessary for the protection of the health and safety of the residents of Bolton and ordered or performed by an administrative agency of the town, state or federal government.
- 1.11.5 Appeal from this Bylaw shall be to the Superior Court within 15 days of the issuance of a decision by the Board of Health.
- 1.11.6 Fine for violation of this Bylaw shall be Fifty Dollars (\$50.00) for each offense, each day of violation to be a separate offense.
- 1.11.7 The invalidity of any section(s) or provision(s) of this Bylaw shall not invalidate any other provision or section thereof.
- 1.11.8 Definitions
 - a) Alter: As used in this Bylaw shall mean the undertaking of any activity or construction which will change the existing drainage patterns, recharge characteristics or recharge, ground or surface water constituents, either temporarily or permanently, or which, through production, storage, or transportation of toxic or hazardous substances in any amounts beyond those necessary for normal household use, may foreseeably result in accidental introduction of those substances into ground or surface waters.
 - b) Hazardous Material: A material or combination of material, which because of its quantity, concentration, or physical, chemical or infectious characteristics may:
 - 1) Cause or significantly contribute to an increase in mortality or an increase in serious, irreversible, or incapacitating illness; or
 - 2) Pose a substantial present or potential hazard to human health or the environment when improperly treated, stored, transported, or disposed of, or otherwise managed.

- c) Maximum Groundwater Elevation: The height of the groundwater table when it is at its maximum level of elevation as stated in State Environmental Code, Title 5.
- d) Person: As used in this Bylaw shall include: Any individual, group of individuals, association, partnership, corporation, company, business, organization, trust estate, the commonwealth or political subdivision thereof, administrative agency, public or quasi-public corporation or body, or any other legal entity or its legal representatives, agents or assigns.

1.12 COUNCIL ON AGING

The Council on Aging shall consist of nine registered voters of the Town and shall be appointed by the Board of Selectmen for terms not to exceed four years for any member. Said terms shall be staggered so that not more than three appointments shall be made in any calendar year. Members can be re-appointed for consecutive terms. Failure to attend seven (7) out of the regularly scheduled ten (10) monthly meetings may result in termination of membership on the Council.

- 1.12.1 The duties of said Council on Aging shall be to:
 - a) identify the total needs of the community's elderly population;
 - b) educate the community and enlist support and participation of all citizens concerning these needs;
 - c) design, promote or implement services to fill these needs, or coordinate present existing services in the community;
 - d) promote and support any other programs which are designed to assist elderly programs in the community.
- 1.12.2 Said Council on Aging shall coordinate with the Commonwealth of Massachusetts Office of Elderly Affairs and shall be cognizant of all state and federal legislation concerning funding, information exchange, and program planning which exists for better community programming for the elderly.
- 1.12.3 Said Council on Aging shall give an annual report to the Board of Selectmen with a copy of that report directed to the Commonwealth of Massachusetts Office of Elderly Affairs.

1.13 STREET NUMBERING

- 1.13.1 All houses, businesses and structures shall properly display at the front thereof, in a position easily observed from the street on a year round basis, the proper number of the building assigned by the Town. Assignment of numbers shall be obtained from the Building Inspector at the time the building permits are approved. Owners of houses, businesses and structures without numbers at the time of passage of this bylaw shall obtain the proper number from the Building Inspector as soon as possible.
- 1.13.2 In cases where a house, business or structure is not visible from the street, or the distance is too great for a reasonable size number to be easily seen from the street, a sign, no larger than twelve (12) inches by twelve (12) inches with numbers not less than three (3) inches in height and one and one quarter (1.25) inches in width, shall be posted at the street end of the driveway, and on the same side of the street, in such a way as to be seen by emergency vehicles approaching from either direction. This sign may be substituted by numbers not less than three (3) inches in height and one and a quarter (1.25) inches in width on both sides of the mailbox only if the mailbox is located at the end of the appropriate driveway and on the same side of the road.
- 1.13.3 In cases where more than one house, business or structure shares a common or shared driveway, or the buildings are not visible from the street, or the distance is too great for a reasonable size number to be easily seen from the street, a sign shall be posted at the street end of the driveway on the same side of the street in such a way as to easily be seen by emergency vehicles approaching from either direction. This sign shall designate the range of numbers for which the driveway serves.

This sign may not exceed the twelve (12) inch by twelve (12) inch dimension in order to accommodate the full set of numbers in a way easily seen from emergency vehicles. The numbers shall not be less than three (3) inches in height and one and a quarter (1.25) inch in width. The owner(s) will be notified by the Building Inspector by letter if this is necessary. Each driveway branching off of the common or shared driveway, and all subsequent branchings, must have additional sign(s), dimensioned and visible as outlined above, at that immediate junction indicating which number(s) are served by that driveway.

- 1.13.4 In all cases each figure shall be at least three (3) inches in height and shall be clearly visible. Said numbers shall contrast with their background. Numbers placed on signs must be placed at a height of not less than three (3) feet from ground level and not exceed five (5) feet from ground level. Signs must be kept clear of brush and other obstacles that would limit visibility of said sign by emergency vehicles.
- 1.13.5 In cases where a house, business or structure remains unnumbered, or where the numbering may have been lost or destroyed as to be illegible, the owner shall

replace the numbering within twenty (20) days after official notification by the Board of Selectmen.

1.13.6 Any violation of Section 1.13.5 beyond the twenty (20) day limit will result in a fine of not more than \$10.00 per day. No permit or Certificate of Compliance of any kind (i.e. occupancy, building, electrical, Smoke Detector Certificate of Compliance) shall be issued by the Town to the owners if their building, dwelling, business or structure is in violation, and does not have a number visible in compliance with this bylaw.

1.14	PERSONNEL BYLAW TABLE OF CONTENTS		
1.14.1	Scope of Bylaw22		
1.14.2	Equal Employment Opportunity22		
1.14.3	Personnel Committee		
1.14.4	Definition of Employee Categories		
1.14.5	90-Day Assessment Period		
1.14.6	Work Week		
1.14.7	Interruption of Service		
1.14.8	Grievance Procedure		
1.14.9	Fringe Benefits.31a)Vacation.b)Sick Leave.c)Personal Leave.d)Bereavement Leave.3233e)Holidays.f)Jury Duty.33g)Military		
1.14.10 a) b) c) d) e)	Compensation35Compensation Review35Overtime/Compensatory Time35Payroll Period36Call-Back Pay36Longevity Pay36		
1.14.11	Amendments to the Bylaw		
1.14.12	Collective Bargaining Agreements		
1.14.13	Severability Clause		
1.14.14	Implementation		
1.14.15	Reimbursement for Town Business Expenses37		
1.14.16	Disciplinary Policy		

1.14.1 Scope of the Bylaw

This Bylaw has been adopted pursuant to the provisions of M.G.L. Chapter 41, sections 108A and 108C, and other enabling acts. This bylaw does not superceed Federal or State Laws and Regulations.

The bylaw applies to all employees of the Town except:

- a) those filled by popular election
- b) those under the control of the School Committee; and
- c) those covered by collective bargaining agreement.

1.14.2 Equal Employment Opportunity

The provisions of this bylaw shall be applied equally to all employees without discrimination as to age, sex, marital status, race, color, creed, national origin, handicap, veteran status, or political affiliation. The Town of Bolton abides by all state and federal employment laws including but not limited to Americans with Disabilities Act (ADA) and Immigration Law Compliance.

a) At-Will Employment

Employees of the Town of Bolton shall serve "at will" and shall be terminated at any time, with or without notice, absent a limiting statue or contractual agreement between the employer and the employee.

b) Equal Opportunity Employer

The Town of Bolton is an equal opportunity employer. All employment decisions are made on a non-discriminatory basis without regard to race, color, national or ethnic origin, sex, marital status or sexual orientation, age, disability, religious or political beliefs, veteran status, or any other factors which cannot lawfully be the basis for an employment decision.

c) Alcohol and Drug Free Workplace Policy

The Town of Bolton hereby notifies employees that the unlawful manufacture, distribution, dispensing, possession, or use of a controlled substance is prohibited, and violation of such prohibition in the workplace can lead to dismissal.

1.14.2.1 Harassment

Harassment is conduct that has the purpose or effect of creating an intimidating, hostile or offensive working environment, interfering with an individual's work performance, or otherwise adversely affecting an individual's employment opportunities.

Derogatory comments and objectionable conduct of a racist, ethnic, or sexist nature or those aimed at a person's handicap are not only abusive and offensive, but are also violations of the laws, policies, and guidelines of equal opportunity. No employee either male or female is to be subjected to unsolicited and unwelcome conduct of a sexual nature, nor is any employee to be subjected to comments which are disparaging to his/her handicap, sex, or racial or ethnic background.

1.14.2.2 Sexual Harassment

Introduction

It is the goal of the Town of Bolton to promote a workplace that is free of sexual harassment. Sexual harassment of employees occurring in the workplace, or in other settings in which employees may find themselves in connection with their employment, is unlawful and will not be tolerated by this town. Further, any retaliation against an individual who has complained about sexual harassment or retaliation against individuals for cooperating with an investigation of sexual harassment complaint is similarly unlawful and will not be tolerated. To achieve our goal of providing a workplace free from sexual harassment, the conduct that is described in this policy will not be tolerated and we have provided a procedure by which inappropriate conduct will be dealt with, if encountered by employees.

Because the Town of Bolton takes allegations of sexual harassment seriously, we will respond promptly to complaints of sexual harassment and where it is determined that such inappropriate conduct has occurred, we will act promptly to eliminate the conduct and impose such corrective action as is necessary, including disciplinary action where appropriate

Please note that while this policy sets forth our goals of promoting a workplace that is free of sexual harassment, the policy is not designed or intended to limit our authority to discipline or take remedial action for workplace conduct which we deem unacceptable, regardless of whether that conduct satisfies the definition of sexual harassment.

Definition of Sexual Harassment

In Massachusetts, the legal definition of sexual harassment is this:

"sexual harassment" means sexual advances, requests for sexual favors, and verbal or physical conduct of a sexual nature when:

submission to or rejection of such advances, requests or conduct is made either explicitly or implicitly a term or condition of employment or as a basis for employment decisions; or,

such advances, requests or conduct have the purpose or effect of unreasonably interfering with an individual's work performance by creating an intimidating, hostile, humiliating or sexually offensive work environment.

Under these definitions, direct or implied requests by a supervisor for sexual favors in exchange for actual or promised job benefits such as favorable reviews, salary increases, promotions, increased benefits, or continued employment constitutes sexual harassment.

The legal definition of sexual harassment is broad and in addition to the above examples, other sexually oriented conduct, whether it is intended or not, that is unwelcome and has the effect of creating a work place environment that is hostile, offensive, intimidating, or humiliating to male or female workers may also constitute sexual harassment.

While it is not possible to list all those additional circumstances that may constitute sexual harassment, the following are some examples of conduct which if unwelcome, may constitute sexual harassment depending upon the totality of the circumstances including the severity of the conduct and its pervasiveness:

> Unwelcome sexual advances - whether they involve physical touching or not, including repeated, offensive sexual flirtations, advances or propositions;

> Sexual epithets, jokes, written or oral references to sexual conduct, gossip regarding one's sex life; comment on an individual's body, comment about an individual's sexual activity, deficiencies, or prowess; Displaying sexually suggestive objects, pictures, cartoons;

> Unwelcome leering, whistling, brushing against the body, sexual gestures, suggestive or insulting comments; and

Discussions of one's sexual activities.

Incidents of sexual harassment may result from the conduct of managers, supervisors, or other employees, and may also result from the conduct of the Town's vendors, clients or suppliers or other parties with whom the Town deals with on a regular basis. All incidents of harassment, whether occurring internally within the Town's offices, or externally when an employee of the Town is performing services at any other location, should be reported promptly to the contact persons identified in the complaint procedure outlined below.

Complaint Procedure

In compliance with Massachusetts General Laws Chapter 151B, Section 3A, the following sexual harassment complaint procedure has been developed by the Personnel Committee specifically to ensure that complaints are investigated quickly and in a manner that is fair to all:

1. Any employee having a complaint of harassment should immediately notify his/her immediate supervisor. If this complaint is against the immediate supervisor or if the employee does not feel comfortable in approaching his/her immediate supervisor relative to the compliant, the employee should immediately notify the Personnel Committee, or a member thereof. To obtain the work addresses and telephone numbers of The Personnel Committee, contact the Town Secretary at 779-2297.

2. If a supervisor receives a complaint of harassment, the supervisor is required to immediately notify the Personnel Committee. When the Personnel Committee receives the complaint we will promptly investigate the allegation in a fair and expeditious manner. The investigation will be conducted in such a way as to maintain confidentiality to the extent practicable under the circumstances. Our investigation will include a private interview with the person filing the complaint and with witnesses. We will also interview the person alleged to have committed sexual harassment. In the event of such an investigation, each employee, supervisor and manager is responsible for cooperating fully with the Personnel Committee.

3. When we have completed our investigation, we will, to the extent appropriate, inform the person filing the complaint and the person alleged to have committed the conduct of the results of that investigation.

If it is determined that inappropriate conduct has occurred, we will act promptly to eliminate the offending conduct, and where it is appropriate we will also impose disciplinary action.

Disciplinary Action

If it is determined that inappropriate conduct has been committed by one of our employees, we will take such action as is appropriate under the circumstances. Such action may range from counseling to termination from employment, and may include such other forms of disciplinary action as we deem appropriate under the circumstances.

No Retaliation for Filing Complaint or for Cooperating in an Investigation No employee, supervisor or manager shall be retaliated or discriminated against in any way for making a complaint of sexual harassment in good faith or for assisting or cooperating in the investigation of such a complaint. Such retaliation or discrimination is unlawful, and shall not be tolerated by the Town.

State and Federal Remedies

In addition to the above, if an employee believes he/she has been subjected to sexual harassment, they may file a formal complaint with either or both of the government agencies set forth below. Using the Personnel Committee complaint process does not prohibit an employee from also filing a complaint with these agencies.

Please note, each of the agencies has a short time period for filing a claim (EEOC - 180 days; MCAD - 6 months).

The United States Equal Employment Opportunity Commission ("EEOC") One Congress Street, 10th Floor Boston, MA 02114 Telephone: (617) 565-3200 FAX: (617) 565-3196

The Massachusetts Commission Against Discrimination ("MCAD") One Ashburton Place, 6th Floor, Room 601 Boston, MA 02108 Telephone: (617) 727-3990 FAX: (617) 720-6053

1.14.2.3 Recruitment and Appointment Policy

The town shall make every effort to attract and employ qualified persons. All department heads and appointing authorities shall be responsible for the recruitment and selection of personnel. All hiring will be performed in an open environment and hiring decisions shall be made on the basis of qualifications, references, and evidence of required position skills. The qualifications,

classification and salary range for positions shall be established in accordance with the classification and compensation plans adopted by the Personnel Committee.

a) Notice of Vacancies

Department heads and the appointing authority shall, upon the identification of a vacancy or on the authorization of a new position, prepare a job vacancy notice. The job vacancy notice shall be based upon the existing job description and include: the job title, major duties of the position, qualifications, salary, a closing date for applications and application instructions. If a job description does not exist a new job description shall be written and submitted to the Personnel Committee for approval prior to a job vacancy notice being completed. Job vacancy forms can be obtained from the Personnel Committee.

b) Posting and Advertisement of Job Vacancy Notices

Advertising should be adequate to ensure that a sufficient number of qualified applicants apply for available vacancies. In all circumstances, department heads and appointing authorities shall ensure that: notices of vacant positions be posted for five (5) business days on the bulletin boards in prominent work locations (including but not limited to Town Hall, Department of Public Works facilities, the Library, the Communication Center, the Fire Station and the Police Station). In addition, job vacancy notices may be placed in local newspapers or equivalent advertising through the internet for a minimum of 5 days, as needed.

c) Applications

All candidates applying for employment in the town of Bolton shall complete a Bolton official employment application in addition to his/her resume and return the application to the appointing authority on of before the date and place indicated in the position announcement. Each applicant shall sign the employment application, and the truth of all statements shall be certified by the applicant's signature. All candidates who complete the employment application accurately and honestly shall be entitled to a fair and equitable review of their qualification

d) Interviewing

The final applicant(s) for a position will be interviewed by the appropriate department head and/or appointing authority.

e) Examinations

The appointing authority may require an examination as one part of the selection process. Examinations may be written, oral, practical, physical or any combination thereof and shall be relevant to the requirements of the position.

f) References

A candidate's former employers, supervisors, and other references may be contacted as part of the selection process. References and other background investigations shall be documented and made part of the applicant's file. All reference checks and investigations shall be completed prior to the offer of employment by the appointing authority.

g) Application Records

The application, documentation of reference checks, and related documents submitted shall be maintained in the employee's personnel file.

h) Offers

All employment offers shall be made in writing by the appointing authority. The written notice of employment shall include the salary, the starting date, and any other appropriate information. Copies of the letter of employment shall be provided to the Personnel Committee.

i) Medical Examination

Some positions may require a medical examination for employment with the town. If so required, the medical examination shall take place after receipt of notice of an employment offer by the appointing authority and prior to the starting date of employment. The examination shall be at the expense of the town by a physician or medical institution selected by the Personnel Committee. The examining physician shall advise as to whether or not, in the opinion of the physician, the applicant is fit to perform the duties of the position for which the applicant has been selected. If, in the opinion of the appointing authority, the position and reasonable accommodation is not possible, the offer of employment will be withdrawn. The appointing authority reserves solely to itself the right to waive physical or medical requirements.

j) Failure to Report

An applicant, who accepts an offer and fails to report to work on the date set by the appointing authority, shall, unless excused by the appointing authority, be deemed to have declined the offer of employment and shall be withdrawn.

k) Department Head Responsibilities

Department heads shall provide and supervise on-site training and orientation regarding specific rules, regulations, policies and procedures of the employee's assigned department including the safety policies and procedures. Department heads during the orientation period of any employee shall at reasonable intervals discuss work performance with the new employee. The department head shall be responsible for documenting these discussions.

l) Conflict of Interest

The Conflict of Interest law prohibits a municipal employee or town official from participating in any particular matter affecting the financial interest of an "immediate family member". The purpose of the broad prohibition against acts

of nepotism is to prevent potential conflicts or the appearance of favoritism, which arise whenever a public official's personal loyalty to a family member competes with the need to make objective personnel decisions. "Immediate family member" is defined in the law as the employee, employee's spouse or partner, their parents, children, brothers, and sisters.

1.14.3 Personnel Committee

This Personnel Bylaw shall be administered by a Personnel Committee, consisting of three voting members appointed by the Board of Selectmen, said Board shall have the power to fill all vacancies. Members are appointed for three-year, staggered terms. The Committee shall select a chairman from its membership who shall preside over meetings.

Like all Town employees and board/committee members, the Personnel Committee is subject to and shall abide by the provisions of the Conflict of Interest Law. M.G.L. Chapter 268A.

No members of the Personnel Committee may be an employee of the Town nor hold Town office, whether appointed or elected. Members of the Personnel Committee shall serve without compensation.

At any meeting of the Personnel Committee action by a majority of the Committee members present shall be binding. At least two members of the Committee shall be present in order to constitute a quorum.

The Personnel Committee is responsible for administering this Personnel Bylaw and may establish procedures as necessary to fulfill this charge. The Personnel Committee may, from time to time, make and issue interpretations and regulations consistent with the provisions of the Personnel Bylaw and necessary for its administration. Similarly, the Personnel Committee shall periodically review the Bylaw and make recommendations for changes, as appropriate, and in accordance with the provisions governing amendments as set forth in Section 1.14.11.

It shall be the responsibility of the Personnel Committee to develop a classification and compensation plan, and to review position classifications and rates of pay at reasonable intervals, as set forth in Section 1.14.10(a). The Personnel Committee shall conduct an open meeting once annually with town employees to discuss employee concerns about salary, fringe benefits, and the administration of the Bylaw. The Personnel Committee shall fulfill its role in the Grievance Procedure, as outlined in Section 1.14.8.

Personnel Records

The Personnel Committee shall be responsible for establishing and maintaining personnel records as may be required by law, and as necessary for effective personnel management. All employees covered under the Personnel bylaw, appointing authorities, and department heads shall comply with and assist in furnishing records, reports and information as may be requested by the Personnel Committee or the person or persons assigned responsibility for personnel administration.

a) Contents of records

The Personnel Committee shall maintain or cause to be maintained an individual personnel file for each employee which may include, but not be limited to, the following: the employment application or resume; a copy of any documented reference checks and background investigation reports; a report of all personnel actions reflecting the original appointment, promotion, demotion, reassignment, transfer, separation, or layoff; history of employment and correspondence directly related to the employee's past employment record; reclassification or change in the employee's rate of pay or position title; commendations, records of disciplinary action, training records, performance evaluation, and other records that may be pertinent to the employee's employment record.

b) Confidentiality and Access to Records

Personnel records shall be considered confidential and access to records shall be limited to the Personnel Committee, persons authorized by the Personnel Committee to administer the personnel system, appointing authorities and department heads. Any employee may upon request to the employee's appointing authority have access to review their personnel file. The employee's review of their employment shall be in the presence of the employee's department head or appointing authority. An employee may also obtain a copy of his/her personnel file upon five (5) days written notice.

c) Location of Records

A central file for all positions under the personnel bylaw shall be established in the office of the Personnel Committee. The Town Secretary will maintain original files for the Personnel Committee. The Department Head, provided the file copies are under lock and key, may keep a copy of the file and access is restricted.

d) Release of Information

Unless written authorization is received from an employee, except to verify employment dates, no information concerning an employee shall be released

- 1.14.4 Definition of Employee Categories
 - a) FULL TIME EMPLOYEE is a regular employee who works at least 40 hours per week throughout the year.
 - b) PART TIME EMPLOYEE is a regular employee who works less than 40 hours per week throughout the year.
 - c) TEMPORARY EMPLOYEE is an employee in a full or part time position which is not likely to require the services of an incumbent on a year-round basis. Seasonal employees and employees hired for a specific project on a short-term basis are considered temporary employees.
 - d) REGULAR EMPLOYEE is an employee who has completed his/her 90-Day Assessment Period.
 - e) EXEMPT EMPLOYEE is a salaried employee who is employed in an executive, administrative, or professional capacity, and is not generally entitled to overtime pay if she/he meets the following criteria:
 - 1) Executive--primary duty is to manage a department;
 - 2) Administrative--primary duty is office or non-manual work directly related to management policies, or directly assisting an executive; and
 - 3) Professional--primary duty requires advanced knowledge acquired by specialized study, work is intellectual and the result is not standardized.
 - f) NON-EXEMPT EMPLOYEE is an employee, whether paid a salary or hourly wage, whose primary duty is not executive, administrative, or professional in nature. A non-exempt employee is entitled to overtime pay under certain conditions.
- 1.14.5 90-Day Assessment Period

The first ninety (90) days of employment between the Town and new employees is considered an assessment period wherein both parties assess overall match of skills, abilities, and job needs to determine if the correct employment decision has been made. Prior to the conclusion of this period either party may dissolve the employment relationship consistent with the prevailing practices of the employment-at-will relationship.

During the 90-Day Assessment Period applicable employee benefits are made available to eligible employees. However, if the employment relationship is dissolved at any time during the 90-Day Assessment Period, exiting employees will not be entitled to any accrued vacation time. An employee is entitled to pay for holidays during the 90-Day Assessment Period, and may use sick leave as earned.

1.14.6 Work Week

The normal work week for employees shall be as follows:Managerial EmployeesAs requiredPolice Employees40 hours per weekHighway Employees40 hours per weekCommunication Employees40 hours per week

1.14.7 Interruption of Service

Any regular employee who leaves Town service while in good standing may be rehired by the appointing authority and, upon successful completion of one year's service, will be given credit for time actually employed by the Town.

1.14.8 Grievance Procedure

A grievance is a dispute between an employee and the appointing or supervisory authority arising out of an exercise of management rights or administrative discretion, or interpretation of this Bylaw.

STEP I. An employee who has a grievance must discuss the grievance with the department head and/or supervisory authority in a mutual effort to resolve the grievance.

STEP 2. If, one week after such conference, a satisfactory understanding and solution of the grievance has not been reached, then either the department head or the employee may refer the grievance to the Personnel Committee. The Committee shall hold a hearing thereon and render a written decision within ten working days of such hearing.

STEP 3. If any party is aggrieved by the decision of the Personnel Committee, such decision may be appealed in writing to the Board of Selectmen, who shall conduct a hearing with respect hereto and render a written decision within ten days of such hearing. A decision of the Board of Selectmen with respect to the grievance shall be binding on both parties.

1.14.9 Fringe Benefits

a) Vacation Time

Full time employees shall be entitled to paid vacation in accordance with the following schedule:

Less than five (5) years of service	two (2) weeks/year
From five (5) years to	
less than eleven (11) years of service	three (3) weeks/year
Upon completion of eleven (11) years of service	four (4) weeks/year
Upon completion of twenty (20) years of service	five (5) weeks/year

An employee shall request vacation leave from his/her supervisor, giving as much notice as possible. An employee is eligible for vacation leave after completing one year of service and may begin taking leave as accrued after that time. If a holiday falls during a vacation period, an additional day off will be scheduled, by agreement with the department head.

Vacation leave is credited monthly, beginning in the first month of employment, at the rate of 1/12 of the employee's annual entitlement. Vacation leave may not be taken until it is earned. Vacation entitlement for the first and last months of an individual's employment shall be pro-rated on

a calendar day basis to the date of hire or termination, as appropriate. For vacation increment purposes, the anniversary date shall be considered the first day of the month of date of hire, for those employees hired between the first and the fifteenth day of the month. Those employees hired after the fifteenth day of the month shall have an anniversary date of the first day of the following month, for vacation increment purposes.

Part time employees working an average of at least 20 hours per week are entitled to vacation according to the above schedule, with their vacation pay pro-rated to their weekly schedule.

Temporary employees or employees working fewer than 20 hours per week are not entitled to vacation pay.

Employees are required to take vacation on a regular basis to allow for the proper rest from the rigors of work. In unusual circumstances and with prior approval of the department head and Personnel Committee, no more than (5) accrued vacation days may be carried into, and must be taken, during the next calendar year.

b) Sick Leave

Each full time employee shall accumulate sick leave at the rate of one and one quarter day for each month of employment (15 days / year). Unused sick leave may be accumulated up to a maximum of 120 working days. Sick leave is intended for use in times of illness only and as such carries no "buy-back" or "cash in" valuation.

New employees are entitled to sick leave after completing at least one month of service.

Part time employees who work a minimum of 20 hours per week are eligible to accumulate sick leave on a pro-rated basis.

Sick leave is generally for protection of employees against loss of pay due to personal illness. However, sick leave may be used by an employee for illness of a dependent child, spouse, or other immediate relative. Medical certification may be required for any illness after five (5) consecutive days.

c) Personal Leave

Each full time employee with at least one year of service is allowed two days each fiscal year for personal reasons. Personal leave is not to be used as vacation, and may not be combined with vacation leave. Except in an emergency, employees should obtain approval from their supervisors at least 48 hours in advance of taking the personal day. Personal leave may not be carried over from one fiscal year to the next.

Part time employees who work a minimum of 20 hours per week are eligible for personal leave pro-rated basis to their weekly schedule.

d) Bereavement Leave

An employee may be granted up to three days of bereavement leave for the purpose of arranging for and attending the funeral of a member of the employee's immediate family. "Immediate family" shall include spouse, child, parent, brother, sister, parent-in-law, children-in-law, or other relative residing in the employee's household. An employee may be granted up to one day of bereavement leave for the purpose of attending the funeral of a grandparent, aunt, uncle, or other close relative.

e) Holidays

Floating one-half day	
New Year's Day	January 1
Martin Luther King Day	3rd Monday in January
Washington's Birthday	3rd Monday in February
Patriots' Day	3rd Monday in April
Memorial Day	Last Monday in May
Independence Day	July 4
Labor Day	1st Monday in September
Columbus Day	2nd Monday in October
Veterans' Day	November 11
Thanksgiving Day	4th Thursday in November
Day after Thanksgiving	4th Friday in November
Christmas	December 25

Full time employees and part time employees who work at least 20 hours per week will receive compensation on a pro-rated basis for the above listed holidays. Employees working on a holiday will be paid double time, with the exception of the December 25th holiday when all working employees will be paid at a rate of double time plus one half time therefore equal to two and one half times their regular pay.

f) Jury Duty

Employees required to serve on a jury shall be paid the difference between the compensation received from jury duty (excluding travel allowance) and their regular compensation from the Town. In most cases, this will be full pay for the first three days of jury duty, and full pay less the daily jury compensation paid by the court system for the balance of the time served on the jury. Proper evidence of jury pay received must be submitted to the department head.

g) Military Leave

If a regular employee who is a member of an organized unit of the ready reserve of the armed forces gives written notice to the Department Head of the date that the employee wishes to depart for the purpose of military training or call to active duty, the employee shall be entitled to the benefits provided by MGL Chapter 149, Section 52A. Such employee shall be paid the compensation that would otherwise have been received during the

leave, less all monies that the employee receives from the military (excluding reimbursement for out-of-pocket expenses).

- h) Deleted December 13, 1999, Article 3
- i) Leaves of Absence

The Town of Bolton abides by all federal and state law regarding family and medical leave, to include, but not limited to Family and Medical Leave Action (FMLA), Massachusetts Maternity Leave Act (MMLA), and Small Necessities Leave Act (SNLA), and other applicable federal and state laws regarding leaves of absence.

If you are in need of a leave of absence consult with the Personnel Committee to review your options.

j) Insurance Benefits

The Town and employee shall share equally in the cost of health and life insurance benefits. The employee's share of the premium payment is deducted from his/her pay check. Employees working a minimum of 20 hours per week on a year-round basis are entitled to join the Town's group insurance program. The Town currently offers health insurance programs through a commercial carrier and Health Maintenance Organizations. The Town also offers a \$5,000 Savings Bank Life Insurance Policy.

Employees who are members of Bolton health insurance group and terminate employment (for reasons other than gross misconduct) may continue coverage in the group for up to 18 months, at their own expense. Such continued coverage will be identical to the coverage provided under the plan for active employees and their covered families, but will cease if the premium is not paid on a timely basis or if they become covered under another group health plan.

Health insurance coverage may be continued, at their own expense, for up to 36 months for a spouse or dependent child of an employee who dies while employed by the Town. In the case of a divorce or legal separation, health insurance may be continued for dependents, at their own expense or at the expense of the employee, in accordance with the divorce/separation agreement. Continuation of this coverage also ceases if the premium is not paid on a timely basis or if coverage is obtained through another group. In addition, dependent children have the right to continue coverage with their own membership, at their own expense, if coverage through the Town's group is lost when they cease to have "dependent child" status under the terms of the health insurance policy.

k) Workers' Compensation/Injury Leave

Police and Fire uniformed employees are covered under the provisions of M.G.L. Chapter 4l Section 100 and 111F, and are entitled to the rights and benefits of these laws. All other employees are covered under the Massachusetts Worker's Compensation Law and are entitled to the benefits and provisions of this law.

l) Pension

Regular employees working a minimum of 20 hours per week (1000 hours per year) are required to join the Worcester County Retirement Association, with payroll deductions made in accordance with State Law and date of hire.

m) Educational Assistance

Full time employees and part time employees working at least 20 hours per week, who have worked for the Town for at least one year are eligible for reimbursement for tuition, registration fees, and books for work-related courses which serve to improve their knowledge and skills and increase their performance with the Town.

Approval for the particular course must be requested prior to enrollment in order to be eligible for the reimbursement, and in order to qualify must have the recommendation of the department head and approval of the Personnel Committee. Approval is subject to sufficient municipal funds. Reimbursement shall be made upon successful completion of the course or program. The Town may require the employee to sign an agreement to remain with the Town for a period of up to two years after completion of the course, or else be willing to reimburse the Town for the funds.

"Successful completion" of a course shall mean a minimum grade of B or equivalent.

1.14.10 Compensation

a) Compensation Review

Whenever the Personnel Committee reviews the wage and salary provisions of this Bylaw, it shall take into account and give such weight as it may deem desirable to the following:

- Rates of pay for like positions in other Massachusetts towns considered by the Personnel Committee to be comparable to Bolton;

- Rates of pay for like jobs (if any) in commercial and business establishments in the area of Bolton and vicinity;

- Other benefits received by Town employees;
- The current level of the Consumer Price Index for the Boston area; and
- The financial policy and economic considerations of the Town.

b) Overtime/Compensatory Time

Payment for overtime shall be in accordance with the terms of the Fair Labor Standards Act of 1938, as amended. If an assignment requires work in excess of 40 hours per week, such overtime work must be authorized in

advance by the department head. Non-exempt employees shall be paid one and one-half times their regular hourly rate for hours worked beyond 40 in the work week. Compensatory time may be taken in lieu of overtime pay, at the rate of time and one-half, only by mutual agreement of both employee and supervisor prior to overtime hours being worked. If such agreement is made, then compensatory time should be taken within a reasonable time of being earned. An employee is eligible to be paid for any unused compensatory time upon termination or retirement. Supervisory, professional, and managerial employees are exempt employees and are not eligible for overtime pay.

c) Payroll Period

Employees are paid on a bi-weekly basis for a period beginning on a Sunday and ending on a Saturday. Payday is generally Wednesday for the two-week period ending the previous Saturday.

d) Call-Back Pay

If full time non-exempt employees are recalled to work from off-duty hours, the Highway Department will receive a minimum of three hours pay, and the Police Department will receive a minimum of three hours pay.

e) Longevity Pay

All full time employees shall receive longevity payments according to the following schedule:

Length of Service	Annual Amount
10 years	\$200
15 years	\$300
20 years	\$400

On an annual basis the Selectmen should review any full-time employee who has accumulated over 20 years of service to the Town and may award a meritorious award up to \$100 for this service.

This amount shall be paid in one lump sum in the first pay period following the anniversary of the employee's date of hire. The date of hire shall be considered the date when the employee began working for the Town at least 20 hours per week on a continuous basis. The years of service shall be calculated from this date of hire.

All part time employees working at least 20 hours per week are entitled to longevity pay according to the above schedule, with the amount of payment pro-rated to their weekly schedule.

Temporary employees, or employees working fewer than 20 hours per week are not entitled to longevity pay.

1.14.11 Amendments to the Bylaw

This Bylaw may be amended by vote of the Town at a Town Meeting. How ever, no amendment shall be considered or voted on by Town Meeting unless the proposed amendment has first been considered by the Personnel Committee and Board of Selectmen.

1.14.12 Collective Bargaining Agreements The provisions of any collective bargaining agreement negotiated, as provided by the Mass. General Laws, between the Town and an employee, group or union shall prevail over the provisions of this Bylaw.

1.14.13 Severability Clauses

Each provision of this Bylaw shall be construed as separate, to the end that if any part of it shall be held invalid for any reason, the remainder shall continue in full force and effect.

1.14.14 Implementation Implementation date of Sections 1.14.9 and 1.14.10 of this bylaw and all their respective subsections shall be implemented and become effective commencing July 1, 1988.

- 1.14.15 Reimbursement for Town Business Expenses
 - (a) Mileage Reimbursement

When employees have prior approval from their supervisor to use their personal vehicle for Town business, they are reimbursed for business travel at the approved per mile rate, set by the Personnel Committee and the Town Treasurer concurrently. Local travel expenses between the employee's home and assigned work location are not reimbursable. However, if employees are required by business necessity to travel from home directly to a site other than their assigned work location, the Town will reimburse them for the difference between the mileage in their normal commute to their assigned work location and the total miles driven on business.

When using a private vehicle for Town business purposes, the employee assumes liability for the vehicle. All employees who use their personal vehicles for Town business must have a current driver's license and vehicle liability insurance in the minimum amounts required by state law. Otherwise, the vehicle is not authorized for Town business use.

1.14.16 Disciplinary Policy

All employees are responsible for observing regulations necessary for proper operation of town departments. Disciplinary actions shall be the responsibility of supervisors, department heads and appointing authorities, who shall exercise their responsibility with discretion and with concern for the employee. Reasons for Disciplinary Action

a)

Disciplinary action may be initiated for failure of an employee to fulfill responsibilities as an employee. Sufficient cause for disciplinary action shall include, but is not intended to be limited to the following:

- Incompetence or inefficiency in performing assigned duties
- Failure to perform a reasonable amount of work, violation of any reasonable official order or failure to carry out any lawful and reasonable directions made by a proper supervisor
- Habitual tardiness or absence from duty.
- Falsification of time sheets or other official documents
- Illegal use or possession of narcotics or alcohol while on duty
- Misuse or unauthorized use of town property
- Fraud in securing appointment
- Disclosure of confidential information
- Abuse of sick leave or absence without leave
- Conviction of a felony
- Violation of safety rules, practices and policies.
- Engaging in sexual harassment
- Any situation or instance of such seriousness that disciplinary action is warranted
- b) Disciplinary Procedures

Department heads and supervisors shall be responsible for enforcing rules and regulations. The type of disciplinary action imposed is at the discretion of appointing authorities and department heads and is dependent upon the nature of the disciplinary violation. Disciplinary action shall include only the following: oral reprimand, written reprimand, suspensions, and discharge.

- c) Oral Reprimand A department head observing action of an employee warranting disciplinary action may issue an oral warning to the employee. The oral warning shall be presented with maximum regard for minimizing embarrassment to the employee and may include a statement concerning the purpose of the warning. An oral reprimand shall be noted in the employee's personnel file.
- d) Written Reprimand The department head may issue a written warning including reasons for the warning and an offer of assistance on the part of the department head in correcting the unsatisfactory situation. A copy of the written warning shall be placed in the employee's personnel file and carry a specified period in which the behavior shall be improved.

38

- (e) Suspension At the discretion of a department head and with sufficient cause a department head may suspend an employee without pay for a period or periods not to exceed twenty (20) working days in any twelve (12) month period. Suspension may be in lieu of oral reprimand, written reprimand, and disciplinary probation and may be effective immediately. Within forty-eight (48) hours of the effective date of the suspension the employee shall be provided with a written notice stating the reasons for the length of the suspension.
- f) Discharge An employee may be discharged for unsatisfactory job performance, violation of the town regulations or after the exhausting of other disciplinary procedures. The department head shall provide the employee with a written notice stating the reason or reasons for the discharge and the effective date of the discharge.
- g) Neither the existence of, nor anything contained within, this policy shall alter in any way the AT-WILL employee status of any employee.

1.15 HOUSING AUTHORITY

1.15.1 Purpose

For the creation, development and management of housing and housing programs in the Town of Bolton.

1.15.2 Membership

There will be 5 members, 4 of whom are elected by the town. The fifth is appointed by the Executive Office of Communities and Development (EOCD). At creation of authority, members will be appointed by the Board of Selectmen, these individuals will serve until members are elected at the next town election. Terms are as follows:

Individual receiving highest # of votes:	5 year term
Individual receiving 2nd highest # of votes:	4 year term
Individual receiving 3rd highest # of votes:	2 year term
Individual receiving 4th highest # of votes:	l year term

The individual appointed by EOCD will serve an initial term of 3 years.

As terms expire, successor will be appointed or elected in same way, but for a term of five years. Membership is restricted to residents of the town. Vacancies, aside from the expiration of terms will be filled/appointed by the Board of Selectmen until new members can be elected at the next town election.

1.16 LICENSES AND PERMITS OF DELINQUENT TAXPAYERS

- 1.16.1 The Tax collector, or other municipal official responsible for records of all municipal taxes, assessments, betterments and other municipal charges, hereinafter referred to as the tax collector, shall annually furnish to each department, board, commission, or division, hereinafter referred to as the licensing authority, that issues licenses or permits including renewals and transfers, a list of any person, corporation, or business enterprise, hereinafter referred to as the party, that has neglected or refused to pay any local taxes, fees, assessments, betterments or other municipal charges for not less than a twelve month period, and that such party has not filed in good faith a pending application for an abatement of such tax or a pending petition before the appellate tax board.
- 1.16.2 The licensing authority may deny, revoke or suspend any license or permit, including renewals and transfers of any party whose name appears on said list furnished to the licensing authority from the tax collector, provided, however, that written notice is given to the party and the tax collector, as required by applicable provisions of law, and the party is given a hearing, to be held not earlier than fourteen days after said notice. Said list shall be prima facie evidence for denial, revocation or suspension of said license or permit to any party. The tax collector shall have the right to intervene in any hearing conducted with respect to such license denial, revocation or suspension. Any

findings made by the licensing authority with respect to such license denial, revocation or suspension shall be made only for the purposes of such proceeding and shall not be relevant to or introduced in any other proceeding at law, except for any appeal from such license denial, revocation or suspension. Any license or permit denied, suspended or revoked under this section shall not be reissued or renewed until the license authority receives a certificate issued by the tax collector that the party is in good standing with respect to any and all local taxes, fees, assessments, betterments or other municipal charges, payable to the municipality as the date of issuance of said certificate.

- 1.16.3 Any party shall be given an opportunity to enter into a payment agreement, thereby allowing the licensing authority to issue a certificate indicating said limitations to the license or permit and the validity of said license shall be conditioned upon the satisfactory compliance with said agreement. Failure to comply with said agreement shall be grounds for the suspension or revocation of said license or permit; provided, however, that the holder be given notice and a hearing as required by applicable provisions of law.
- 1.16.4 The Board of Selectmen may waive such denial, suspension or revocation if it finds there is no direct or indirect business interest by the property owner, its officers or stockholders, if any, or members of his immediate family, as defined in section one of chapter two hundred and sixty eight in the business or activity conducted in or on said property.
- 1.16.5 This section shall not apply to the following licenses and permits: open burning, S 13 of C 48; bicycle permits, S 11A of C 85; sales of articles for charitable purposes, S 33 of C 101; children work permits, S 69 of C 149; clubs, associations dispensing food or beverage licenses, S 21E of C 140; dog licenses, S 137 C 140; fishing hunting, trapping licenses, S 12 of Ch 131; marriage licenses, S 28 of C 207; theatrical events, public exhibition permits S 181 C 140.

1.17 DEMOLITION DELAY BYLAW

- 1.17.1 Intent and Purposes The purpose of this bylaw is to establish a procedure for reviewing requests to demolish significant buildings in order to preserve, when possible, the historical, cultural and architectural heritage of Bolton.
- 1.17.2 Definitions
- 1.17.2.1 Building- a structure capable of being used to shelter persons, animals, materials or equipment.
- 1.17.2.2 Commission- the Bolton Historical Commission

- 1.17.2.3 Demolition- any voluntary act of pulling down, destroying, burning, removing, or razing of a building in whole or in part; commencing the work of destruction; or allowing the same to be done by others.
- 1.17.2.4 Applicant- person or persons filing an application for review under this bylaw. If the person submitting the application is not the owner, the owner will be required to indicate his/her assent to the filing of the notice or application.
- 1.17.2.5 Significant building- any building which is historical by reason of its age (in whole or in part seventy-five (75) or more years) *and* is determined by the Commission to be either (1) associated with a person or event contributing to the cultural, political, economic, social or architectural history of the town, commonwealth or the United States of America; or (2) historically or architecturally important (in terms of period, style, construction, or association with an architect or builder), either by itself or in the context of a group of buildings.
- 1.17.2.6 Preferably preserved significant building- any significant building for which it is determined after hearing to be in the public's interest to preserve or rehabilitate rather than demolish under this bylaw.
- 1.17.2.7 Demolition permit- any permit (demolition, alteration, building) issued by the Building Inspector, as required by the State Building Code, which authorizes the demolition of a building (excluding interior demolition).
- 1.17.2.8 Building Inspector- Bolton Inspector of Buildings or person(s) otherwise authorized to issue demolition permits.
- 1.17.2.9 Day- any calendar day, including Saturdays, Sundays and holidays.
- 1.17.3 Procedure
- 1.17.3.1 No demolition permit shall be issued for a significant building without first fully complying with the provisions of this bylaw.
- 1.17.3.2 A written demolition permit application shall be filed with the Building Inspector.
- 1.17.3.3 If the demolition permit application is for a building which is seventy-five (75) or more years old, the building's historical significance will be determined by the Commission within fourteen (14) days of the Commission's receipt of the application. The Commission will notify the Building Inspector and applicant in writing of this initial determination. If the building is not determined to be significant, the Building Inspector may issue a permit to demolish in accordance with all applicable procedures.
- 1.17.3.4 If the building is determined to be significant, the Commission shall hold a public hearing within thirty (30) days of this initial determination to further

determine whether the demolition of the building will be detrimental to the historical, cultural or architectural heritage of the town. Notice of the time, place and purpose of the hearing shall be given twice in a local newspaper, the first notice at least fourteen (14) days before and the second notice at least seven (7) days before such hearing and by posting a notice in the Town Hall for a period of at least fourteen (14) days before such hearing and by mailing a notice of hearing to the applicant.

- 1.17.3.5 A determination shall be made by the Commission within seven (7) days of the close of the public hearing. The applicant, Building Inspector and the Town Clerk shall be provided with a copy of the determination.
- 1.17.3.6 If the Commission determines the demolition of the building in question is not detrimental to the historical, cultural or architectural heritage of the town, the applicant may apply for the necessary permits to begin work.
- 1.17.3.7 If the Commission determines the demolition of the building in question is detrimental to the historical, cultural or architectural heritage of the town, the building shall be deemed a preferably preserved significant building and no further demolition permits may be applied for or issued for a period of six (6) months from the date of such determination.
- 1.17.3.8 During the six (6) month period, the Commission will invite the applicant (and owner of record, if different from applicant) to participate in an investigation of alternatives to demolition. If acceptable alternatives are agreed upon by the Commission and the applicant, the Commission will file a copy of said agreement with the Building Inspector and Town Clerk and the applicant may apply for necessary permits to begin work. Work shall only be done in accordance with the terms of the agreement unless and until new permit applications are filed and processed hereunder.
- 1.17.3.9. If the Commission is satisfied that there is no feasible alternative to demolition, the Commission may so advise the applicant, Building Inspector and Town Clerk in writing, at any time during this six (6) month period, and the Building Inspector may issue a permit to demolish in accordance with all applicable procedures.
- 1.17.4 Emergency Demolitions
- 1.17.4.1 Nothing in this bylaw restricts the demolition of a significant building determined by the Building Inspector to present a danger to public safety which only demolition can prevent. Such a determination shall be made in accordance with the applicable provisions of the State Building Code and after consultation with the Commission Chairperson or designee. Whenever an emergency demolition permit is issued under the provisions of this section, a written report must be filed with the Commission describing the condition of the building and the basis for the determination.

- 1.17.5 Enforcement and Remedies
- 1.17.5.1 The Commission and the Building Inspector are each authorized to enforce the provisions of this bylaw.
- 1.17.5.2 The Building Inspector shall not issue a permit pertaining to any property on which a significant building has been demolished voluntarily in whole or in part without first fully complying with the provisions of this bylaw for a period of two (2) years from the date of demolition.
- 1.17.5.3 The provisions of this bylaw may be enforced by a non-criminal disposition pursuant to Section 1.1.5 of the Administrative Bylaws of the Town of Bolton.
- 1.17.6 Administration
- 1.17.6.1 The Commission may adopt such rules and regulations as are necessary to administer the terms of this bylaw.
- 1.17.6.2 The Commission is authorized to adopt a schedule of fees to cover the costs associated with the administration and review of any application which is filed under this bylaw.
- 1.17.7 Severability
- 1.17.7.1 If any provision of this bylaw is determined to be invalid or unconstitutional by any court, every other section shall continue in full force and effect."
- 1.18 WETLANDS BYLAW
- 1.18.1 Purpose

The purpose of this Bylaw is to conserve and protect the resource areas, the resource interests, and natural resource services, in the Town of Bolton by regulating activities deemed by the Conservation Commission ("Commission") likely to have a significant or cumulative adverse effect upon resource interests. Protected resource interests include, but are not limited to: flood control, storm damage prevention, public and private water supplies, ground water, water quality, prevention of pollution and sedimentation, fisheries, wildlife habitat, rare species habitat including rare plant species, agriculture, aquaculture, adjoining land areas and recreation deemed important to the community. Natural resource services may be classified as either ecological service --the physical, chemical, or biological functions that one resource provides for another; or public service -- the public uses of natural resources or functions of natural resources that benefit the public. This Bylaw is intended to utilize the Home Rule authority of the Town of Bolton to conserve and protect additional resource areas, with additional standards and procedures stricter than those of the Wetlands Protection Act, (M.G.L. Ch. 131, Section 40) and Regulations thereunder, (310 CMR 10.00).

1.18.2 Jurisdiction

Unless excepted in Section 1.18.3 or pursuant to a Wetlands Bylaw Permit, no person shall commence to remove, fill, dredge, build upon, degrade, pollute, discharge into, or otherwise alter the following resource areas: any freshwater wetlands; marshes; wet meadows; bogs; swamps; vernal pool habitat, including but not limited to state certified vernal pools; banks; fresh water seeps; reservoirs; lakes; rivers; streams and creeks, whether perennial or intermittent; riverfront areas which are (a) lands within two hundred feet (200') of perennial rivers or streams; beaches; lands under water bodies; lands subject to flooding or inundation by ground water or surface water; and land within twenty-five feet (25') of the above resources areas, except for riverfront areas (collectively the "wetland resource areas" protected under this bylaw) and adjacent upland resource areas (collectively the "adjacent upland resource areas" protected This 25-foot prohibition, however, shall not apply to under this bylaw). crossings essential to access upland areas. Said resource areas shall be protected whether or not they border surface waters.

Adjacent upland resource areas shall include all lands within seventy-five feet (75') of wetland resource areas enumerated above, except for perennial streams and rivers for which the adjacent upland resource area extends for two hundred feet (200') from the top of the bank.

Except as expressly permitted by the Commission or as provided in this Bylaw, no person shall commence to remove, fill, dredge, build upon, degrade, pollute, discharge into, or otherwise alter a wetland resource area, or an adjacent upland area, as described above.

Where a proposed activity involves work within a resource area, the Commission shall presume that the area is significant to protect the resource interests, enumerated in Section 1.18.1.

Where the proposed activity involves work within the riverfront area, the Commission shall presume the area is significant to protect the riverfront area and its resource interests. This 200-foot presumption does not apply to the redevelopment of those portions of riverfront areas regarded as "previously developed" or "degraded" or to "paths". Orders of Conditions for redevelopment and paths are granted at the discretion of the Commission.

These presumptions are rebuttable and may be overcome by a preponderance of the evidence showing that the resource area does not play a role in the protection of one or more of these interests. In the event that the presumption is deemed to have been overcome as to the protection of all the resource interests, the Commission shall make a written determination to this effect, setting forth its grounds.

Where the applicant provides information that the resource area at the site of activity does not play a role in the protection of an interest, the Commission may determine that the presumption for that interest has been rebutted. Where the applicant provides information that site of the activity plays a partial role in the protection of an interest, the Commission may determine that the presumption for that interest has been partially rebutted and the presumption of significance is partially overcome.

1.18.3 Applicability, Limitations, and Exceptions

Where the presumption set forth in Section 1.18.2 is not overcome, the applicant shall prove by a preponderance of the evidence that there are no practicable and substantially equivalent economic alternatives to the proposed project with less adverse effects on the interests identified in Section 1.18.1. Further, the applicant shall prove that the work including proposed mitigation will have no significant adverse or cumulative adverse effect on the resource areas or resource interests. In the event that the Commission finds that the applicant has failed to make either of said proofs, it shall make a written determination setting forth its grounds in an Order of Conditions that shall impose conditions that will protect the interests which make the resource area significant or shall in a written determination deny the activity as it cannot be conditioned to protect the interests of the Bylaw and/or its regulations.

To prevent the loss of resource areas, applicants shall be required to avoid, where feasible, altering a resource area; minimize alteration of a resource area; and, where alteration is unavoidable, complete full mitigation. Replication of resource areas may be required as a form of mitigation.

Exceptions to the Bylaw shall be limited to: maintaining, repairing or replacing, adding to, but not substantially changing or enlarging, an existing single-family residential structure, septic system or appurtenance; maintaining landscaping and gardens accessory to an existing single-family residential structure; lands lawfully in agriculture (commercial and non-commercial) at the time the work takes place; forest cutting (as defined in 310 CMR 10.04, 'Agriculture'(b) 14); maintaining or repairing, but not substantially changing or enlarging, an existing structure in a resource area, such as drainage structures, culverts, bridges, driveways or roadways; maintaining or repairing, but not substantially changing or enlarging, fire protection water holes, artificial ponds; clearing of water courses, conservation and outdoor recreation; existing orders and filings before the Commission prior to the effective date of this Bylaw; public utilities (as defined in 310 CMR part 10.53 section (3)(d); projects carried out under the direction of the U.S. Natural Resource Conservation Service; and emergency projects necessary for the protection of the health and safety of the public and subject to the provisions and conditions of 310 CMR part 10.06.

No activities other than the excepted activities shall commence without receiving and complying with a permit issued pursuant to this Bylaw. No work proposed in any Wetlands Bylaw Permit application shall be undertaken until the Wetlands Bylaw Permit issued has been recorded in the registry of deeds, or if the land is registered land, filed in the appropriate land court, and until the holder of the Wetlands Bylaw Permit certifies in writing that the Wetlands Bylaw Permit has been recorded. Such certification shall include the book and page or instrument number and date.

1.18.4 Applications for Wetlands Bylaw Permit and Requests for Determination Any person desiring to know whether a proposed activity is excepted or an area is subject to this Bylaw may request a determination from the Commission by filing a Request for Determination of Applicability (RFD). Such a Request for Determination of Applicability shall include information and plans as required by the Commission.

Written application shall be filed with the Commission to perform activities within the resource areas protected by this Bylaw. The Wetlands Bylaw Permit application shall include such information and plans as deemed necessary by the Commission to describe proposed activities and their effects on the resources areas. Where appropriate, the Commission may accept the Notice of Intent and plans filed under the Wetlands Protection Act and the Regulations as the Wetlands Bylaw Permit application and plans under this Bylaw.

At the time of a request for determination or a Wetlands Bylaw Permit application, the applicant shall pay a filing fee specified in the Regulations of the Commission. The fee is in addition to that required by the Wetlands Protection Act (G.L. Ch 131 Section 40) and Regulations (310 CMR 10.00). The fee shall be deposited in a dedicated account, for use only for wetland protection activities. Town, county, state and federal projects are exempt from the filing fee.

The Commission is authorized to require the applicant to pay for the reasonable costs and expenses borne by the Commission for specific expert engineering or for other outside consultant services in order to reach a final decision on the application.

The Commission may require that the applicant's performance and observance of the Order of Conditions, including mitigation, be secured wholly or in part by one or more of the methods set forth in the regulations. This security shall be in addition to any security required by any other town or state board, agency or official.

1.18.5 Notice and Hearings

Wetlands Bylaw Permit applications shall be filed with the Commission subject to the provisions and conditions of 310 CMR part 10.00 and the Commission may, in an appropriate case continue a public hearing for good cause and may combine its hearing under this Bylaw with the hearing conducted pursuant to the Wetlands Protection Act, M.G.L. Ch 131, Section 40 and Regulations, 310 CMR 10.00.

1.18.6 Wetlands Bylaw Permits, and Certificates of Compliance

Within 21 days of the close of the hearing the Commission shall issue or deny a Wetlands Bylaw Permit for the activities requested. If a Wetlands Bylaw Permit is issued, the Commission shall impose conditions, which the Commission deems necessary or desirable to protect resource areas, resource interests and natural resource services, and all activities shall be done in accordance with those conditions.

Due consideration shall be given to any demonstrated hardship on the applicant by reason of denial, as presented at the public hearing.

Wetlands Bylaw Permits issued hereunder shall expire three years from the date of issuance and may be renewed by the applicant for additional one year periods only where a written request for renewal is received by the Commission not less than 30 days prior to the expiration of the Wetlands Bylaw Permit and that good cause has been shown for said extension and that there is no likely significant or cumulative adverse effect upon any of the resource area or resource interests.

Notwithstanding the above, a Wetlands Bylaw Permit may contain requirements which shall be enforceable for a stated number of years, indefinitely, or until permanent protection is in place, and shall apply to all owners of the land.

If the activity has been completed in accordance with said Wetlands Bylaw Permit, the Commission shall, within 21 days after a request, issue a Certificate of Compliance evidencing such determination, which may be combined with the Certificate of Compliance under the Wetlands Protection Act. A Certificate of Compliance may specify conditions, which will continue to apply for a fixed number of years or permanently and shall apply to all owners of the land. The Certificate of Compliance shall be recorded in the Registry of Deeds, or if the land is registered land, in the appropriate land court, and the Commission notified in writing by the holder of the Wetlands Bylaw Permit that the Certificate of Compliance has been recorded. Such notification shall include the book and page or instrument number and date.

Violations of this Bylaw, submission of false information or new information that substantially alters the likely impact of the project on the resource areas may cause the Commission to revoke or modify a Wetlands Bylaw Permit or determination issued under this Bylaw after notice to the holder of the Wetlands Bylaw Permit or determination, notice to the public, abutters and town boards, pursuant to Section 1.18.5 and a public hearing.

Appeal from this Bylaw shall be to the Superior Court.

1.18.7 Regulations

After public notice and public hearing, the Commission may promulgate rules and regulations to effectuate the purpose of this Bylaw effective when voted and filed with the Town Clerk. Failure to promulgate such rules and regulations or a legal declaration of their invalidity by court of law shall not act to suspend or invalidate the effect of this Bylaw.

1.18.8 Definitions

Except as otherwise provided in this Bylaw or in regulations of the Commission, the definitions of terms in this Bylaw shall be as set forth in the Wetlands Protection Act, M.G.L. Ch 131 Section 40, and Regulations 310 CMR 10.00.

1.18.9 Enforcement

The Commission shall have the authority to enforce this Bylaw, its regulations, and Wetlands Bylaw Permit issued thereunder by violation notices, administrative orders, and civil and criminal court actions. Any person who violates provisions of this Bylaw, its regulations and/or Wetlands Bylaw Permits may be ordered to restore the property to its original condition, pay damages and take other action deemed necessary to remedy such violations, or may be fined, or both.

Any person, who violates any provisions of this Bylaw or regulations, Wetlands Bylaw Permits or administrative orders issued thereunder, may be served with a Notice of Violation enumerating the alleged violations. As an alternative to criminal prosecution, the Commission may elect to utilize the non-criminal disposition procedure set forth in M.G.L. Ch 40, Section 21D.

The violator shall pay any and all costs including reasonable attorney fees incurred by the Town. The fine for a violation of this Bylaw shall be Three Hundred Dollars (\$300.00) for each offense; each day of the violation shall be a separate offense. Enforcement will be done in accordance with Step Enforcement Policy Against Alleged Violations of the Wetlands Protection Act and other-Local By-laws under the jurisdiction of the Bolton Conservation Commission, dated May 2, 2005.

Municipal boards and officers, including any police officer or other officer having police powers, shall have authority to assist the Commission in enforcement.

1.18.10 Severability

The invalidity of any section or provision of this Bylaw shall not invalidate any other section or provision thereof, nor shall it invalidate any Wetlands Bylaw Permit or determination, which previously has been issued.

1.19 DISCHARGE OF SNOW OR ICE ONTO PUBLIC WAYS

No person other than an employee in the service of the Town of Bolton, or an employee in the service of an independent contractor acting for the Town of Bolton, shall pile, push, or blow snow or ice onto a public way that is plowed and sanded by the Town.

Whoever violates this provision shall be punished by a fine of one hundred dollars (\$100). This provision is to be enforced by the Police Department.

1.20 TOWN ADMINISTRATOR

1.20.1 The Board of Selectmen may appoint a Town Administrator for a three-year term in accordance with Mass. Gen. Laws Chapter 41, Section 23A. The Town Administrator shall be appointed on the basis of executive and administrative qualifications and shall be especially fitted by education, training and previous experience in public administration to perform the duties of the office. The Town Administrator, under the direction and supervision of the Board of Selectmen, shall be responsible for handling the day-to-day administrative affairs of the Town and for implementing the policies set forth by the Board of Selectmen. The Town Administrator shall act for the Board of Selectmen in any matter that it may assign to the Town Administrator. With the approval of the Board of Selectmen, the Town Administrator may perform other duties as may be requested by any other Town Officer, Board, Committee or Commission. The administrator shall be responsible for coordinating and implementing general office operating policies that shall apply to all elected and non-elected employees as may be agreed upon by the Board of Selectmen and other boards. The Town Administrator shall receive such compensation for all services performed as the Board of Selectmen shall determine; said compensation shall not exceed the amount appropriated therefore by the Town. The Board of Selectmen may remove the Town Administrator at its discretion.

1.21 AGRICULTURAL PRESERVATION BYLAW

1.21.1 Legislative Purpose and Intent

The purpose and intent of this Bylaw is to state with emphasis the Right to Farm granted to all citizens of the Commonwealth under Article 97 of the Constitution, and all state statutes and regulations thereunder including but not limited to Massachusetts General Laws Chapter 40A, Section 3, Paragraph 1; Chapter 90, Section 9; Chapter 111, Section 125A; and Chapter 128 Section 1A. We the citizens of Bolton restate and republish these rights pursuant to the Town's authority conferred by Article 89 of the Articles of Amendment of the Massachusetts Constitution, ("Home Rule Amendment").

This General Bylaw encourages the pursuit of agriculture, promotes agriculturebased economic opportunities, and protects farmlands within the Town of Bolton by allowing agricultural uses and related activities to function with minimal conflict with abutters and Town agencies. This Bylaw shall apply to all jurisdictional areas within the Town.

1.21.2 Definitions

The word "FARM" shall include any parcel or contiguous parcels of land, or water bodies used for the purpose of agriculture, or accessory thereto.

The words "AGRICULTURE" or "FARMING" or their derivatives shall include, but not be limited to the following:

- farming in all its branches and the cultivation and tillage of the soil;
- dairying;
- production, cultivation, growing, and harvesting of any agricultural, aquacultural, floricultural, viticultural, or horticultural commodities;
- growing and harvesting of forest products upon forest land, and any other forestry or lumbering operations;
- 4-H or The National FFA Organization projects; and
- keeping and/or raising of livestock which includes but is not limited to horses, poultry, swine, cattle, sheep, goats, ratites (such as emus, ostriches and rheas), camelids (such as llamas and alpacas), and other domesticated animals, including bees and fur-bearing animals

"FARMING" shall encompass activities including, but not limited to, the following:

- operation and transportation of slow-moving farm equipment over roads within the Town;
- control of pests, including, but not limited to, insects, weeds, predators and disease organism of plants and animals;
- application of manure, fertilizers and pesticides;
- conducting agriculture-related educational and farm-based recreational activities, including agri-tourism, provided that the

activities are related to marketing the agricultural output or services of the farm;

- processing and packaging of the agricultural output of the farm and the operation of a farmer's market or farm stand including signage thereto;
- maintenance, repair, or storage of seasonal equipment, or apparatus owned or leased by the farm owner or manager used expressly for the purpose of propagation, processing, management, or sale of the agricultural products; and
- on-farm relocation of earth and the clearing of ground for farming operations.

1.21.3 Right To Farm Declaration

The Right to Farm is hereby recognized to exist within the Town of Bolton. The above-described agricultural activities may occur on holidays, weekdays, and weekends by night or day and shall include the attendant incidental noise, odors, dust, and fumes associated with normally accepted agricultural practices. It is hereby determined that whatever impact may be caused to others through the normal practice of agriculture is more than offset by the benefits of farming to the neighborhood, community, and society in general. The benefits and protections of this Bylaw are intended to apply to those agricultural and farming operations and activities conducted in accordance with generally accepted agricultural practices. Moreover, nothing in this Agricultural Preservation Bylaw shall be deemed as acquiring any interest in land, or as imposing any land use regulation, which is properly the subject of state statute, regulation, or local zoning law.

1.21.4 Agricultural Commission

For the furtherance of the goals of this Bylaw, there is hereby established an Agricultural Commission, to consist of five (5) members, appointed by the Board of Selectmen, at least three (3) of whom shall be representative of the Bolton farming community and one (1) from the non-farming, residential community. The term of appointment shall be for three (3) years, timed to coincide with the Town's fiscal year. At the time said Commission is first appointed, one member shall be appointed for one (1) year, two members for a term of two (2) years, and two members for a term of three (3) years; all subsequent appointments shall be made for terms of three (3) years. The appointing authority shall fill any occurring vacancy based on the unexpired term being vacated.

The Agricultural Commission shall be authorized:

- To act as a spokesperson for the Bolton farming community;
- To present written or oral testimony in the context of public hearings before any Town Board or Committee;
- On its own initiative, to advise any Town Board or Committee on matters pertinent to the particular area of jurisdiction of that Board or Committee, insofar as the issue relates to the interests of agriculture;

- To advise the Board of Selectmen and/or any other Town Board on any Town-wide issues relating to agriculture which, in the opinion of the Agricultural Commission, require a concerted Town action or response;
- To respond to any request for information or advice from any Town Board or Committee;
- To provide public information or public education services regarding agriculture and the practice of agriculture in Bolton, or in general;
- To provide education and information for farmers in residential areas on how to minimize their impact on their neighbors;
- To serve, at the written request of any Town Officer, Board, or Committee, as an arbitrator or negotiator in the resolution of disputes relating to agricultural issues, provided that no order or decision on the particular matter in question has already been issued by any Town entity legally authorized to do so; and that any resulting decision or recommendation shall be strictly advisory, and shall have no legal standing or definitive nature beyond the voluntary agreement of the parties to abide by it;
- To promote farming business and agriculture in Town, to create awareness through education, and the promotion of agriculture and its benefits to Bolton.

The Agricultural Commission, *per se*, is specifically not authorized:

- To acquire or hold property, real or personal;
- To act as an agent or representative of any individual or entity in any matter pending before any Town or State agency;
- To interfere with, litigate, or serve in any way as a conduit, agent, or forum for appeals relating to any decision made by a legally authorized Town Officer or agency.
- 1.21.5 Disclosure Notification

Right to Farm Notification

"It is the policy of this community to conserve, protect and encourage the maintenance and improvement of agricultural land for the production of food, and other agricultural products, and also for its natural and ecological value. This disclosure notification is to inform residents that Bolton is a Right to Farm community where farming activities occur. Such farming activities may include, but are not limited to, activities that cause noise, dust and odors. Residents and property owners are also informed that the location of their property within the Town may be impacted by agricultural operations including the ability to access water services for such property under certain circumstances."

A copy of this disclosure notification shall be posted by the Town to residents and property owners each fiscal year by mail in the Town census and in one or more of the following forms: annual report, official Town website, transfer station, or library.

1.21.6 Resolution of Disputes

Any person who seeks to complain about the operation of a farm may, notwithstanding pursuing any other available remedy, file a grievance with the Board of Selectmen, the Zoning Enforcement Officer, or the Board of Health, depending upon the nature of the grievance. The filing of the grievance does not suspend the time within which to pursue any other available remedies that the aggrieved may have. The Zoning Enforcement Officer or Board of Selectmen may forward a copy of the grievance to the Agricultural Commission or its agent, which shall review and facilitate the resolution of the grievance, and report its recommendations to the referring Town authority within an agreed upon timeframe.

The Board of Health, except in cases of imminent danger or public health risk, may forward a copy of the grievance to the Agricultural Commission or its agent, which shall review and facilitate the resolution of the grievance, and report its recommendations to the Board of Health within an agreed upon timeframe.

1.21.7 Severability

In the event that any part of this Bylaw is, for any reason, held to be unconstitutional, invalid, or void by any court or agency of competent jurisdiction, it shall not be construed to affect the validity or the continuation and implementation of all other elements of this Bylaw.

2.0 ZONING BYLAWS

2.1 PURPOSE AND ADMINISTRATION

2.1.1 Purpose

This Zoning Bylaw is enacted pursuant to, and under the authority of, Chapter 40A of the General Laws, and amendments thereto, and of Article 89 of the Amendments to the Constitution, for the purpose of guiding the sound development of the Town of Bolton.

2.1.2 Administration

2.1.2.1 Compliance Certification

No "development" shall be undertaken without certification by the Inspector of Buildings that such action is in compliance with the then applicable Zoning Bylaw, or without review by him regarding whether all necessary permits have been received from those governmental agencies from which approval is required by federal, state, or local law. "Development" for these purposes shall mean erecting, moving, substantially altering a building, sign, or other structure or constructing a driveway. Issuance of a Building Permit or Certificate of Use and Occupancy, where required under the Commonwealth of Massachusetts State Building Code, may serve as such certification.

2.1.2.2 Zoning Review

Materials submitted in application for such certification shall upon receipt be reviewed and forwarded by the Inspector of Buildings to the Board of Selectmen, which within 14 days of receipt by it shall determine whether or not the proposal is in compliance with all Bylaws and is otherwise eligible to proceed and shall so notify the Inspector of Buildings.

2.1.2.3 Enforcement

The Selectmen shall take all such action as may be necessary to secure full compliance with the provisions of the Bylaw, permits issued thereunder and variances granted under the Bylaw or applicable statutes.

2.1.2.4 Penalties

The penalty for violation of any provision of this Bylaw, or any of the conditions under which a permit is issued, or any decision rendered by the Board of Appeals shall be Fifty Dollars (\$50.00) for each offense. Each day that each violation continues shall constitute a separate offense.

The preceding paragraph notwithstanding, any person violating the provisions of any section of these zoning bylaws or condition to special permits and variances which is subject to a specific penalty may be penalized by a non-criminal disposition as provided in M.G.L. Chapter 40 Section 21D which includes the giving, to the violators, a written notice to appear before the clerk of the district court.

2.1.3 Applicability

2.1.3.1 Other Laws

Where the application of this Bylaw imposes greater restrictions than those imposed by any other regulations, permits, restrictions, easements, covenants or agreements, the provisions of this Bylaw shall control.

2.1.3.2 Conformance

Construction or operations under a Building or Special Permit shall conform to any subsequent amendment of this Bylaw unless the use or construction is commenced within a period of six months after the issuance of the permit and in cases involving construction, unless such construction is continued through to completion as continuously and expeditiously as is reasonable.

2.1.3.3 Pre-existing Nonconforming Uses and Structures

(a) Extension or Alteration: As provided in Section 6 of Chapter 40A, G.L., a pre-existing nonconforming single or two-family dwelling may be altered or extended provided that doing so does not increase the nonconforming nature of said structure, and other pre-existing nonconforming structures or use may be extended or altered, on Special Permit from the Board of Appeals if the Board of Appeals finds that such extension or alteration will not be substantially more detrimental to the neighborhood than the existing nonconforming structures or use. Generally an increase of not over 100% from the original floor area of the building at the time of the adoption of the Zoning Bylaw on April 13, 1972, or not over 50% of the ground area in use at that time, shall be considered as not substantially more detrimental to the neighborhood. Once changed to a conforming use, no structure or land shall be permitted to revert to a nonconforming use.

(b) <u>Restoration</u>: Any legally pre-existing nonconforming building or structure may be reconstructed if destroyed by accidental or natural cause if reconstruction is begun within twelve months and completed within twenty-four months from the date of the catastrophe (unless a longer period is allowed through a Special Permit), or else such reconstruction must comply with this Bylaw.

(c) <u>Abandonment</u>: A pre-existing nonconforming use which has been abandoned, discontinued for a period of two years, or changed to a conforming use, shall not be re-established, and any future use of the premises shall conform with this Bylaw.

2.1.3.4 Nonconforming Lots ("The 1973 Amendment")

This section shall not prohibit the erection of a single-family residence on a lot or parcel of land located within the residential zone of the Town of Bolton as set forth in the Base Map if said lot or parcel does not meet the minimum requirements of this amendment provided however that, prior to the effective date in 1958 of the original Bylaw, the lot or parcel of land was in single or joint ownership by a deed and/or a plan recorded in the Worcester District Registry of Deeds or that the lot or parcel of land was in single or joint ownership by deed recorded in the Worcester District Registry or a plan recorded in the same registry between the effective date in 1958 of the original Bylaw and the effective date of 1973 amendment of this section, and that the plan or plans, as recorded, bear the proper approvals of the Planning Board of the Town of Bolton where required by law; and said deed or plans otherwise conform to the Bylaws existent between the effective date in 1958 of the original Bylaw and the effective date of the 1973 amendment, for use as a single family residential building lot.

2.1.4 Validity

The invalidity of any section or provisions of the Bylaw shall not invalidate any other section or provision hereof.

2.2 BOARD OF APPEALS

2.2.1 Establishment

A Board of Appeals of five members shall be appointed by the Selectmen in accordance with the provisions of Chapter 40A of the General Laws, as amended. The Board of Selectmen shall also appoint in like manner two associate members of the Board of Appeals; and in case of a temporary vacancy, absence, inability to act, or conflict of interest on the part of a member of said Board, his place may be taken by an associate member designated by the Chairman or Vice Chairman of the Board of Appeals.

2.2.2 Powers

The Board of Appeals shall have and exercise all the powers granted to it by Chapters 23B, 40A, 40B and 4l of the General Laws, and by this Bylaw, those powers being to hear and decide applications for Special Permits upon which the Board is empowered to act under this Bylaw; to hear and decide petitions for variance, excluding variances for use, to hear and decide other appeals from any aggrieved person, officer, or board; to issue comprehensive permits as provided by Sections 20-23, Chapter 40B, G.L.; in special cases to issue withheld building permits as provided by Section 81Y, Chapter 4l, G.L.; and to act on appeals under the Commonwealth of Massachusetts State Building Code, as provided by Section 23 of Chapter 23B, G.L.

2.2.3 Special Permits

2.2.3.1 Special Permit Granting Authority

Special Permit applications shall be heard and decided upon by the Board of Appeals, except in the case where some other Special Permit Granting Authority is specified by this Bylaw.

2.2.3.2 Criteria

Special Permits provided for in the Zoning Bylaws shall be granted only upon determination by the Special Permit Granting Authority that the activity may be carried out without substantial detriment to the public good and without nullifying or substantially derogating from the intent or purpose of this Bylaw, upon consideration of environmental, economic, fiscal, traffic, public facility, visual, and social consequences. Such permits may also impose conditions, safeguards and limitations on time or use.

2.2.3.3 Expiration

Special Permits shall lapse twenty-four (24) months following grant thereof (excepting such time required to pursue or await the determination of an appeal referred to in Section 17, Chapter 40A, G.L.) if a substantial use or construction has not sooner commenced except for good cause.

2.2.3.4 Public Hearing

Special permits or any extension, modification, or renewal thereof shall only be issued following public hearings held within sixty-five (65) days after filing a complete application with the Town Clerk, a copy of which shall forthwith be filed by the applicant with the Special Permit Granting Authority (SPGA) or its agent. The decision of the SPGA regarding a special permit or any extension, modification, or renewal thereof, shall be made within ninety (90) days following the date of the close of the public hearing. Failure by the SPGA to take final action within said ninety (90) days or extended time, if applicable, shall be deemed to be a grant of the special permit in accordance with, and subject to, the requirements of the twelfth paragraph of Massachusetts General Laws, chapter 40A, section 9 (MGL c. 40A, s. 9). The required time limits for a public hearing and said action may be extended by written agreement between the applicant and SPGA, filed with the town clerk. Said decision shall be filed in writing at the Town Clerk's office within fourteen (14) days of the date the decision is made, and notice of the decision shall forthwith be mailed to the petitioner, applicant, or appellant, and to the parties in interest designated in MGL c. 40A, s. 11, and to every person present at the hearing who so requested and who stated the address to which said notice was to be sent.

2.2.3.5 Planning Board Associate Member

A majority of the joint members of the Board of Selectman and Planning Board may appoint an associate member of the Planning Board. Such associate member, when designated by the chairman of the Planning Board, may serve on the Planning Board for the purpose of acting upon a special permit application, in case of absence, inability to act, or conflict of interest on the part of any member of the Planning Board or in the event of a vacancy on the Board. Said associate member shall be appointed for a term of two years.

2.2.3.6 Rules and Regulations

Such Special Permit Granting Authority may adopt and from time to time amend rules and regulations and charge a fee relative to the issuance of such permits, and shall file a copy of said rules in the Office of the Town Clerk. Such rules may prescribe the application, size, form, contents, style and number of copies of plans and specifications and the procedure for submission and approval of such permits.

2.3 ZONING DISTRICT REGULATIONS

2.3.1 Location of Districts

Zoning Districts are hereby established as shown, located, defined and bounded on a map entitled Base Map and accepted by the Town at the Annual Town Meeting of May 4, 1992, signed by a majority of the Planning Board and filed with the Office of the Town Clerk; which map, together with explanatory matter thereon and all amendments made under this Bylaw, is hereby incorporated in and made a part of this Bylaw including the map referred to in the "Lot Frontage" definition.

2.3.2 Types of Districts and Permitted Uses

No Building or structure shall be erected or used except as set forth in this section or as exempted by Section 2.1.3.4, or by statute. The uses set forth in the following Districts are to be considered exclusive to the said District and are not to be considered mutual to any other District than that in which the allowed use appears. The use of land within the Town for disposal of solid waste derived from premises other than those of the owner or occupant is prohibited except for such operations as may be conducted by the Town of Bolton.

In all Districts uses which are municipal, rural, agricultural, conservation and/or open space are specifically permitted in addition to the uses set forth below:

- (a) Flood Plain: as defined in Section 2.5.3 of the Bylaws of the Town of Bolton
- (b) Residential: for single family residential uses.
- (c) Business: for retail and service establishments where business is conducted wholly within an enclosed building.
- (d) Limited Business: for retail and service establishments subject to Special Permit as required.
- (e) Commercial: for uses set forth in (c) above, for removal or processing of earth materials to include stones, soil, loam, sand or gravel, and for retail, wholesale and service establishments where business is conducted wholly or partially outside a building, except restaurants which must be conducted wholly within an enclosed building.
- (f) Industrial: for storage areas, office buildings, research and development laboratories, light manufacturing industries and compatible allied uses.
- (g) Repealed Annual Town Meeting May 2006
- (h) Limited Recreation Business District, as defined in Section 2.3.8 of the Bylaws of the Town of Bolton.

2.3.3 Mixed Uses No business, commercial, or industrial building or outdoor storage (except for agriculture, horticulture, or floriculture) shall be established on a lot on which a dwelling exists.

2.3.4 The following schedule is included as a general reference schedule of permitted uses within the various districts, but in the event of any inconsistencies, the provisions of 2.3.2 and other provisions of this Bylaw shall be controlling.

In the schedule, "Yes" means a permitted use, "No" means a prohibited use, and "SP" means a use allowed on Special Permit acted on by the Planning Board pursuant to 2.5.5.5. "SSP" means a use allowed on Special Permit acted on by the Board of Selectmen.

- * Except "yes" for buildings erected prior to March 2, 1970
- ** See definition
- *** Except those listed in 2.3.8.4 and 2.3.8.5

**** Only with a 2.5.2.5 Agricultural/Business Use Special Permit

All uses in the Flood Plain District are subject to the provisions of this Bylaw specifically relating to such district.

When an activity might be classified under more than one of the uses, the more specific category shall govern; if equally specific, the more restrictive shall govern.

TYPES OF USES

Limited

Limited				T · ·/ 1					
				Limited					
		Residen	tial	Recreation Business Business			Commercial		Industrial
Business		Residen	liai	Business Business		Comm	lereiai	muusutai	
Rural									
Agriculture, horticul-									
ture,floriculture		Yes	Yes	Yes		Yes		Yes	
	Yes								
Forestry, lumbering,	Yes	Yes	Yes		Yes		Yes		Yes
Open space, conserva-									
tion		Yes	Yes	Yes		Yes		Yes	
	Yes								
Residential									
Single-family dwelling	Yes	No	No		No		No		No
Accessory apartment	Yes	No	No		No		No		No
New two-family dwelling	No	No	No		No		No		No
Conversion to two-		NT Y	NT	NT		NT		N	
family dwelling	Na	No*	No	No		No		No	
Multi family dwalling	No No	No	No		No		No		No
Multi-family dwelling Mobile home** (see	INO	INO	INO		INO		INO		INO
Section 2.5.1)		Yes	No	No		No		No	
Section 2.5.1)	No	105	INU	INU		140		NU	
Mobile home park**	No	No	No		No		No		No
Assisted Living Housing	No	No	SP	No	110	No	110	No	110
Congregate Living Housing	No No	SP	No	110	No	110	No	110	
Nursing Care Facilities	No	No	SP	No		No		No	
Mercantile									
Restaurant (no drive-through									
service)			No	SP	SP	Yes		No	
	No								
Motel, Hotel		No	SP	SP	Yes		No		No
Business-occupied									
trailer		No	No	No		No		No	
	No								
Office building		No	Yes	SP	Yes		Yes		No
Other retail, wholesale or serv		N .T	CD	CD	N 7		N.T.		N.T.
	nin building	No	SP	SP	Yes		No		No
wholly or p	building		No	No	CD.	Yes		No	
outside a	No		INO	INO	SP	res		No	
Manufacturing & Processing	INU								
Light manufacturing									
(see 2.5.5)			No	No	No		No		Yes
	No		110	110	110		110		105
Research & Development									
Laboratories		No	No	No		No		Yes	
	No								
Earth Removal (see Town									
Bylaw 1.10)		No	No	No		Yes		No	
	No								
Bulk storage**		No	No	No		Yes		Yes	
.	No								
Industry-occupied trailer	No	No	No		No		No		No

Limited Recreation Residential Business **Business** Commercial Industrial **Business** Institutional Educational uses exempt from prohibition under S.3 Ch. 40A, G.L. Yes Yes Yes Yes Yes Yes Religious uses Yes Yes Yes Yes Yes Yes Library, Museums, Hospitals Yes Yes Yes Yes Yes Yes Fraternal, Civic, Charitable organizations No Yes Yes Yes Yes Yes Non-municipal solid waste disposal No No No No No No Other municipal uses Yes Yes Yes Yes Yes Yes Recreational Indoor tennis, health club, bowling SP No SP Yes No No Extensive outdoor recreation** Golf course Yes Yes Yes Yes Yes No Other No No No Yes No No Yes Yes Yes Yes Yes No Municipally operated Commercially operated No No No Yes No No Primary Golf Course (see 2.3.8.5) No No No No No Yes **Trail Recreation** (see 2.3.8.4) No No No No No Yes Wireless Communication Wireless Communication Facility/ SSP**** SSP Tower/Communication Devise SSP SSP SSP No Accessory Agricultural/Business Use (see section 2.5.2.5) SSP SSP SSP SSP SSP SSP Roadside produce stands Yes Yes Yes Yes Yes Yes Home occupation (see section 2.5.2.1) Yes Yes Yes Yes Yes Yes Other customary accessory uses Yes Yes Yes Yes Yes No*** Primary Golf Course Accessory Uses (see 2.3.8.5) No No No No No SSP Trail Recreation accessory uses (see 2.3.8.4) No No No No No SSP

Limited

Wireless Communication

Wireless Communication						
Facility/Towers/						
Communication Devise	SSP****	SSP	SSP	SSP	SSP	No

2.3.5 Dimensional Regulations

2.3.5.1 General

No structure shall be erected or extended, or lot created or reduced in size, except through a public taking, unless in conformity with the requirements of this section, or unless specifically exempted by statute or this Bylaw.

2.3.5.2 Dimensional Schedule.

	Resider Lot	ntial		* Backland ot (Residentia	al)	Other Uses i.e. Business, Industrial or Commercial
Minimum lot area**	80,000 sq. ft.		4.5 acres		1.5 acre	es
Minimum lot frontage**	200 feet	50 feet		200 fee	et	
Minimum depth:						
Front yard**	50 feet		50 feet		150 feet	t
Other yards*	*20 feet		50 feet		50 feet	
Minimum lot width for 10 back from street line		2.3.5.5	(b)	150 fee	t	
Maximum lot coverage**	N/A		N/A	see No	tes A and	В

** See definition

*** Refer to 2.3.5.5

- Note A. Maximum lot coverage shall be 8% and the sum of all impermeable surfaces and other areas subjected to vehicle traffic or parking on a regular basis shall not exceed 50% of the total lot area within the zone of proposed use.
- Note B. The definition of Lot Area set forth in Section 2.6, Definitions, shall apply, except that, in Limited Business Zones only, when a landowner has donated to the Town an easement in, or the Town has taken or purchased an easement in, any portion of a tract of land and said easement has been accepted by the Town for the relocation of an already existing public way, said portion of the tract of land shall be included in the calculation of the lot area of the tract for purposes of this bylaw, subject to the following requirements: the easement area is no greater than five (5) percent of the size of the tract; the tract remains the same dimensionally, and is not subdivided, divided, altered, or changed in any way; the boundary lines of said easement area is not included in any calculations of setback requirements; and any use or structure complies with all other requirements of this bylaw.
- 2.3.5.3 One Building Per Lot
 - (a) Residential Use: Not more than a single dwelling shall be erected or moved on a lot.

(b) Other Uses: Not more than a single principal building, regardless of use, shall be erected or moved onto a lot.

- 2.3.5.4 Dimensional Nonconformancy
 - (a) (Deleted 5/2/88 Art 9)

(b) Two-dwelling lots. A single lot containing two or more lawfully existing dwellings less than forty (40) feet apart at the closest point of measurement may be divided into separate lots

despite not meeting yard requirements, provided all other applicable provisions of this Bylaw are satisfied.

In Residential Districts backland or reduced frontage lots may be created by Special Permit issued by the Planning Board. Each such lot must meet the following criteria:

(a) A minimum of street frontage on an accepted and/or approved town way of 50 feet.

(b) The minimum allowable lot width between the dwelling site and the access to the public way is 25'. All plans for such lots shall show and indicate what the minimum lot width is in said plan though it may be in excess of 25'.

(c) The minimum lot size shall be 4 1/2 acres.

(d) Not more than one other Backland lot with frontage contiguous to it will be allowed. A Backland lot or lots as allowed under this Backland Zoning Bylawshall be separated from any other Backland lot or lots by not less than two hundred feet (200') of frontage on an accepted and/or approved right-of-way.

(e) No structure shall be erected within 50 feet of any lot line.

(f) Once a Backland lot is created and approved by the Planning Board, it cannot be subsequently divided to less than four and one-half (4 1/2) acres.

(g) At least 1 l/2 acres of contiguous land which is not in a:

(1) Flood Plain District (Section 2.5.3)

(2) Wetland Resource Areas as defined

in the Wetlands Bylaw 1.18.2

(h) Granting of a Special Permit under this Bylaw does not constitute a waiver of any other applicable Bylaw or statute.

2.3.5.6 Building Height

This bylaw is intended to limit new commercial, business or industrial buildings to a maximum of two stories, plus a basement, with either a flat or a pitched roof.

All buildings, except residential and agricultural uses, shall meet the following height limitations:

(a) The surface of the top occupiable floor, including mezzanines and balconies, shall be no more than fifteen (15) feet above the average ground elevation. Occupiable space as defined and as set forth in the State Building Code shall be the definitive standard of this bylaw.

(b) The highest portion of the roof, excluding spires, steeples, chimneys and antenna, shall be no more than thirty-two (32) feet above the average ground elevation.

The average ground elevation to be used in the above calculations shall be the lower of the average ground elevation on the street side of the building, or the average ground elevation on all sides of the building. In the case of built-up land, the ground elevations prior to such change in contour shall be used.

2.3.5.7 Lot Shape

a. The shape of all lots, except for the exceptions listed below, shall conform to the following formula:

 $16(A) / P^2 > 0.5$ (Sixteen times the area in square feet divided by the square of the perimeter in feet must be greater than one-half) where:

A = Area of the lot in square feet; and, P = Perimeter of the lot in feet.

b. Exceptions:

1. This requirement shall not apply to lots greater than four and one-half (4.5) acres in size.

2. This requirement shall not apply to backland lots created in accordance with Section 2.3.5.5, Backland Zoning.

3. The Planning Board may waive this requirement if it determines that a less stringent requirement will result in a better potential house siting, less environmental damage to the site, or better land use.

2.3.6 Farmland And Open Space Planned Residential Development

2.3.6.1 A Farmland and Open Space Planned Residential Development (FOSPRD) can be created for a residential development by using a special permit.

2.3.6.2 Purpose:

(a) To preserve agriculture and open space

(b) To preserve the rural and historic character of Bolton

(c) To provide land owners with an alternative to traditional subdivisions that allows houses to be built in a way that is more in harmony with local site conditions, without increasing the density of new developments.

2.3.6.3 Prerequisites:

The entire FOSPRD must be in the Town of Bolton, in a residential zone. It must be at least 10 acres, or within fifty (50) feet of another FOSPRD.

2.3.6.4 Maximum Number of Dwellings:

(a) The maximum number of dwellings which can be built in a FOSPRD is equal to the number of dwellings which, is demonstrated to the satisfaction of the Planning Board, could be built on the land in a traditional subdivision.

(b) The applicant shall propose a number of dwellings to the Planning Board by submitting a preliminary plan showing how a traditional subdivision might be laid out, including: roads, lots, house locations and well and septic locations.

(c) Evidence must be submitted to prove that the lots shown would satisfy Board of Health requirements for sewage treatment. This evidence must include a description of soil types and positive percolation tests. The percolation tests shall be done for all the proposed lots, but the Planning Board may waive testing of certain areas if the Planning Board decides that the local soil conditions are suitable for development.

2.3.6.5 Restricted Areas:

(a) Size of Restricted Areas

At least 33% of the area of the FOSPRD must be restricted with either a Farmland Restriction (FR) or an Open Space Restriction (OSR). The 33% requirement may be reached with FR land, OSR land, or some of each. The two types of restrictions are defined below in Sections C & D. No more than 25% of the minimum amount of restricted land can be a Wetland Resource Area as defined in the Wetlands Bylaw 1.18.2 or in a Flood Plain District, as defined in Sections 2.5.3.

(b) How Restricted Areas Are Selected

1. The Planning Board, after consultation with the applicant, Selectmen, Conservation Commission, Board of Health, abutters and general public, shall determine: location of FR land, location of OSR land, location of active use land within OSR land, location and amount of any footpaths. Active use is defined to include ballfields and playgrounds. The Planning Board shall balance the interests of conservation of natural resources, the value of land and existing buildings, and any demonstrated hardship, financial or otherwise that would affect the applicant in determining the location of the restricted area.

2. The intent of this bylaw is for the Planning Board to give preference to FR over OSR, for land that is being farmed or is good farming soils.

3. The required open space shall be configured to the extent practicable, so that it shall be aggregated into a contiguous area, or abut existing significant open space, or achieve one or more of the purposes described in Section 2.3.6.2.

c) Farmland Restriction (FR)

1. Only agricultural or horticultural uses are allowed in the FR area. These uses are defined in Mass General Laws Chapter 61A.

2. No buildings are allowed in the FR area, except 5% of the area can be covered with barns and other accessory farming buildings.

3. The FR land can be owned by either: the Town of Bolton, a conservation trust, a homeowners association of the residents of the FOSPRD, or other owner, such as a farmer. In no event may the FR in any manner be combined, included or joined with or considered as part of the individual dwelling lot area(s). No portion of the FR may be included in determining the minimum dwelling lot area(s) as prescribed in 2.3.5 or 2.3.6.7 a) 1). In all cases the land must be subject to an agricultural preservation restrictions under Mass General Law Chapter 184 s 31 & 32 and that the provisions of that restriction must be reviewed and approved by the Planning Board. The farming rights to the land owned by the Town of Bolton may be sold or leased to farmers.

4. If agricultural activities cease for 3 or more years, then the Town may mow to save any fields or orchards. Fifteen (15) years after the agricultural activity ceases, the Planning Board may, after public hearing and review, cause the FR to be converted to OSR land.

5. No septic systems are allowed in the FR area other than to serve any buildings mentioned in item (c2) above.

6. All five of the above restrictions must be written into the deed of the land in the FR area.

(d) Open Space Restriction (OSR):

l. No new buildings may be constructed on OSR land. No dwellings are allowed in the OSR area.

2. OSR land can be owned by: the Town of Bolton, a conservation trust, or a homeowners association of the residents of the FOSPRD. The restrictions are owned by the Town of Bolton. In no event may the OSR in any manner be combined, included or joined with or considered as part of the individual dwelling lot area(s). No portion of the OSR may be included in determining the minimum dwelling lot area(s) as prescribed in 2.3.5 or 2.3.6.7 a) 1).

3. The OSR land can be used for passive recreation or active recreation. Active recreational uses can not cover more than 10% of the OSR land.

- 4. The above 3 restrictions must be written into the deed of the land in the OSR area.
- 2.3.6.6 Roadways:

(a) When constructing new right-of-way, the applicant must file a subdivision plan and meet all the normal requirements, not withstanding the other provisions of the zoning bylaw:

1. Street widths will be controlled by the Town of Bolton's Subdivision Rules and Regulations and in no case will be less than 18 feet in width.

2. Common driveways serving up to 5 (five) houses each may be obtained under 2.4.1.2 even if none of the lots are backland lots.

2.3.6.7 Dimensional Requirements:

(a) Must meet all dimensional requirements of the Bolton Zoning Bylaws (Section 2.3.5) except:

- 1. Minimum lot area reduced to 1 acre
- 2. Minimum lot frontage reduced to 25 feet
- 3. No "Minimum lot width at 100 feet back from the street" requirement
- 2.3.6.8 Approval Process:

(a) The Planning Board shall be the special permit granting authority for a FOSPRD. The applicant must comply with all the other zoning bylaws of the Town of Bolton.

(b) The Planning Board may issue regulations to go with this bylaw, which specify the application procedure and may charge a fee to cover engineering and legal review and other expenses.

(c) When considering whether or not to grant a special permit, the Planning Board should consider:

1. Whether the proposed plan restricts development on significant farms, farm soils, fields, orchards, hilltops, scenic vistas, stone walls, and other areas important to Bolton's rural character.

2. Whether the proposed plan results in the preservation of buildings with historic significance, which includes buildings more than 100 years old.

3. Whether the proposed plan serves the intent of this bylaw as described in Section 2.3.6.2 better than the alternative of a traditional subdivision.

2.3.6.9 Submission Requirements

The applicant shall submit to the Planning Board copies of the following:

- (a) A traditional Preliminary or Definitive Plan meeting the informational requirements of the Bolton Subdivision Rules and Regulations: and
- (b) A FOSPRD Preliminary or Definitive Plan meeting the informational requirements of the Bolton Subdivision Rules and Regulations

2.3.6.10 Relations to Subdivision Regulations:

In so far as a FOSPRD constitutes a subdivision (per M.G.L. C41, Section 81-L), both the subdivision and the special permitting approval processes may run concurrently. However, subsequent approval by the Planning Board of such portions of the development which constitutes a subdivision shall be required as set forth in the Bolton Subdivision Rules and Regulations, including the approval of the streets and utility system. A favorable action which may be made by the Board on a special permit application for a FOSPRD shall not, therefore, be deemed either to constitute subdivision approval under Bolton Subdivision Rules and Regulations, nor imply that such approval will be given.

2.3.7 Major Residential Development

2.3.7.1 Applicability

Lands affected by this section 2.3.7 includes residentially zoned property, or set of contiguous properties, in common ownership, or in common ownership as of June 10, 1996. The term "common ownership" shall be defined as ownership by the same person or persons or legal entities, or ownership by any two or more persons or entities, as evidenced by control, pursuant to established law.

A Major Residential Development shall mean any division of land which in any five year period:

- a) comprises 15 acres or greater of new residential lots; and
- b) results in the creation of six or more residential lots.

Any subsequent division of land which, if approved, would result in total development in excess of any of the preceding thresholds in any five year period, shall itself be subject to this section 2.3.7, and shall cause any previously approved division(s) of land which contributed to the aforesaid threshold calculation to then also be subject to this section 2.3.7 as a part of that subsequent proposal to divide land.

2.3.7.2 Approval

Major Residential Development shall be allowed only through the special permit process set forth in Section 2.3.6 Farmland and Open Space Planned Residential Development and this section.

2.3.7.3 Purpose

The purpose of this section is to preserve the natural and cultural resources of Bolton by insuring the larger-scale conversion of land to residential use does not consume all of the town's woodlands, fields, farmlands, historic structures and landscapes, orchards, cart paths, rock walls, geologic formations, water courses, wetlands, riparian zones, groundwater recharge areas, hilltops, scenic vistas, and other significant open spaces.

2.3.7.4 Decision

The Planning Board shall make a decision for all Major Residential Developments whether the plan developed in accordance with Section 2.3.6 provides a superior alternative in consideration of the criteria set out in Section 2.3.7.5 in which case, the Planning Board may require that the development occur in accordance with the plan consistent with Section 2.3.6. If the Planning Board does not find that the plan consistent with Section 2.3.6 is superior, the proponent may select whether to pursue development under Section 2.3.6 or use other development options pursuant to Town Bylaws.

2.3.7.5 Decision Criteria

The Planning Board, in making its decision as to whether to require development in accordance with Section 2.3.6 or to allow a traditional division of land, shall evaluate both plans and choose the development method which best meets, in the opinion of the Planning Board, the purpose of this section and the criteria set forth in Section 2.3.6.8 (c).

2.3.8. Limited Recreation Business District

2.3.8.1 This district is intended to promote recreational businesses, to conserve and secure the public health and safety, to facilitate the adequate provision of water and water supply, aquifers and recharge areas, and to protect natural and landscaped conditions as well as open space for commercial use, education, recreation, and the general welfare. The Special Permit Granting Authority for this district shall be the Board of Selectmen. Uses which are municipal, rural, agricultural, conservation and/or open space are permitted as of right without site plan review or special permit.

2.3.8.2 Acreage Required for District

Parcel(s) must be comprised of 80 or more contiguous acres when Trail Recreation is the principal use and 120 or more contiguous acres and an average density of six (6) acres or greater per hole when a Primary Golf Course is the principal use.

2.3.8.3 Primary Uses and Structures allowed as of Right, Subject to Site Plan Approval Processa) Trail Recreation including cross country ski trails, horseback riding trails, or trails used for similar non-motorized recreation, and/or

b) A Primary Golf Course with a minimum of eighteen (18) holes and at least 5000 yards in length, including the use of golf carts.

A site plan shall be submitted in accordance with Section 2.5.5.6 with respect to the above uses.

2.3.8.4 Trail Recreation Accessory Uses and Structures The following Accessory Uses and Structures may be allowed by Special Permit when Trail Recreation is the Principal Use:

a) Function rooms, conference rooms, sale of related equipment, changing and shower rooms, snack bars, offices related to the business of the recreation area.

b) Horse and storage barns, indoor and outdoor riding rings, and maintenance shops.

- 2.3.8.5 Primary Golf Course Accessory Uses and Structures The following Accessory Uses and Structures may be allowed by Special Permit when a Primary Golf Course is the Principal Use:
 - a) Additional golf courses measuring more than 2000 yards.

b) Function rooms, conference rooms, sale of related equipment, health clubs, changing and shower rooms, offices related to the business of the recreation area, restaurants and snack bars. Overnight lodging rooms are allowed at the density of 20 double-occupancy rooms for every 100 acres in the parcel(s) applying for a Special Permit.

c) Garages, cabanas, on-course structures, maintenance shops.

d) Practice putting greens and driving ranges.

e) Sporting uses, including structures, such as tennis, badminton, archery, bocci, racquet ball and shuffle board courts.

f) Other sporting uses such as softball, baseball, basketball and soccer fields with non-permanent bleachers only.

g) Indoor or outdoor swimming pools not exceeding Olympic size.

h) Cross-country ski trails, horseback riding trails or trails used for similar non-motorized recreation, including uses and structures as allowed under Section 2.3.8.4.

i) Other uses and structures determined by the Board of Selectmen to be customarily incidental to the Primary Use and in harmony with the intent and purpose of this bylaw 2.3.8. The applicant shall have the burden to prove by substantial evidence for Selectmen to make a finding that a use or structure is customarily incidental to the Primary Use.

2.3.8.6 Uses and Structures Disallowed include, but are not limited to, the following:

- a) Petting zoos
- b) Miniature golf course
- c) Indoor bowling
- d) Outdoor firing ranges
- e) Heliports or other aircraft landing areas

f) Water parks, e.g. commercial water recreation establishment with slides, wave pools or similar equipment

- g) Camping facilities
- h) Sports stadiums
- i) Casinos or gambling facilities

- j) Snowmobiles
- k) New residential structures
- 1) Motorized trail and/or racing vehicles

2.3.8.7 Definition of Uses & Structures

The definition of "Accessory Uses & Structures" in Bolton bylaw 2.6 shall not apply. For purpose of this bylaw (2.3.8 Limited Recreation Business District) accessory use or structures is a use or structure customarily incidental to and located on the same contiguous land area as the Primary Uses and Structures, and lot coverage requirements complies with Section 2.3.8.8 of this bylaw.

2.3.8.8 Requirements for Special Permit for Accessory Uses & Structures

A. Structures and Impermeable Surfaces

Structures (structures include buildings, sheds, garages, and roofed storage areas, heated or unheated) can occupy not more than three quarters of one percent (0.75%) of total area.

The sum of all impermeable surfaces (including structures, swimming pools, permanent sporting facilities, roads, paved trails and parking) shall not exceed twelve percent (12%) of the total area.

"Section 2.3.5.2 Note A" of Bolton bylaws shall not apply.

B. Structures

1. Bolton bylaw 2.3.5.3 (b), limiting principal structures to one per lot, shall not apply. The Selectmen may authorize more than one principal structure on a lot. An application for said authorization will be considered:

- a) if adequate access is assured for each structure and
- b) if the structure complies with Bolton bylaw 2.5.5.4, fire ponds, and
- c) if the structure complies with Bolton bylaw 2.3.5.6, structure height and uses comply with bylaw 2.5.5.2 activity regulations.

2. Section 2.5.5.3 (b) of the Bolton bylaw shall not apply. No structures shall be located within three hundred (300) feet of any property boundary or town right-of-way and shall be either naturally wooded or visually screened using densely planted evergreen trees large enough to reach five feet (5') in height in three years and of a type which will continue to grow in height. All equipment used in connection with the maintenance of the areas shall be stored in an enclosed structure.

C. Parking

1. Parking must be shielded from the public ways, drainage from parking must not pollute the groundwater, no single parking area shall contain more than sixty (60) parking spaces without a separation of landscaped islands, including shade trees, and the visibility of parking areas shall be limited through structure placements, grading, and landscaping but in no case shall parking areas be closer than two hundred (200) feet to a residential zone border or town right-of-way.

- a) Number of parking spaces will be guided by Bolton bylaw 2.4.1.3, Parking.
- b) Parking requirements may be reduced upon showing reasonable expected traffic volume and adequacy of provided spaces without any overflow into the town's roads.
- c) Bolton bylaw 2.4.1.3 b) shall not apply where the applicant prefers to provide grassed parking.
- d) Night lighting of parking areas shall not be visible at ground level from adjacent properties and shall be constructed with full cut-off fixtures.

2.3.8.9 Driveways

Bolton bylaw 2.4.1.1 a) regulating driveways shall not apply. Instead, the Board of Selectmen may require that shared egress onto town roads be arranged for in order to avoid driveways within two hundred fifty (250) feet of each other or of a street intersection.

2.3.8.10 Protection of Earth Materials

No earth materials shall be removed from the property.

2.3.8.11 Protection of Groundwater

A plan will be submitted for approval as part of the Site Plan that protects groundwater from contamination by fertilizers, herbicides, and pesticides.

2.3.8.12 Special Events

Special Permit may include a provision that Special Events may be permitted. Special Events, defined as a planned event(s) with a total expected attendance of 500 or greater persons using the golf course, trails, and/or facilities, may be permitted upon the issuance of written approval by the Selectmen. Written approval may be granted after recommendations from appropriate Town Boards. These Boards may include (but are not necessarily limited to) the Planning Board, Board of Health, Fire Chief, Building Inspector, and Police Chief. Approval shall consider, but not be limited to, the following criteria:

- i. provisions for adequate off-street parking,
- ii. provision for adequate wastewater disposal,
- iii. provision for traffic control measures for roads leading to the facilities,

- iv. provision for adequate refuse disposal,
- v. provision for crowd control and security and
- vi. provision for adequate fire protection.

Notice shall be given a minimum of forty-five (45) days in advance and the Selectmen will respond and/or call a hearing within twenty-one (21) days. An approval for a special event may be issued only after a finding that there would not be an adverse impact upon residents of the Town of Bolton, including consideration of cumulative impacts.

2.3.8.13 Rules, Regulations and Fees

The Selectmen may write rules and regulations relative to the issuance of special permits under this bylaw and may set fees to cover the expenses of administration. Selectmen may also request the applicant to pay for any special studies identified during the public hearings as needed for traffic control, groundwater protection, and other issues that may substantially impact the resources of the town.

2.3.8.14 Severability

Each provision of this bylaw shall be construed as separate, to the end that if any part of it shall be held invalid for any reason, the remainder shall continue in full force and effect.

2.4 GENERAL REGULATIONS

2.4.1 Driveways and Parking

Overview:

(a) In residential districts, single and shared driveways (driveways serving 2 lots) will be created by permit issued jointly by the Planning Board and the Building Inspector. Upon review of the completed Application for Driveway Permit, the Planning Board will make a determination if a public meeting is required between the Applicant and the Planning Board/Building Inspector to review the submitted driveway plan. Common Driveways (serving 3 to 5 lots) will continue to be administered by the Planning Board per Section 2.4.1.2 of the Bolton Zoning Bylaws.

2.4.1.1 Driveways

(a) Entrances and/or driveways to a given tract of land must be made from an accepted or an approved right-of-way within the extremities of the frontage required except that not more than two single-family dwellings or lots in the Residential Zoning District may maintain one Shared driveway provided that the driveway falls within the frontage of one or both of the lots, and all other provisions of this Bylaw are met.

(b) The slope of the driveway shall not be greater than ten (10) percent within twenty (20) feet of the edge of the street pavement. There shall be no hazardous or blind driveways.

(c) No person shall construct or maintain any driveway, conduit, or drain, so as to discharge water or filth upon the street pavement or into open waterways or ponds.

(d) No driveway shall be constructed which ties directly onto an accepted town way which in any manner inhibits the existing drainage system of the town way.

(e) No single dwelling driveway or Shared Driveway's centerline will be located within thirty five (35) feet of the centerline of any approved or constructed Common Driveway.

(f) The Applicant will be required to complete a Town of Bolton Application for Driveway Permit.

(g) The Planning Board shall issue regulations to go with this bylaw, including charging a filing fee adequate to cover legal review and construction inspection.

2.4.1.2 Common Driveways Serving 3 to 5 Houses

In residential districts, common driveways to serve 3, 4 or 5 lots may be created by Special Permit issued by the Planning Board. Each such common driveway must meet the following criteria:

(a) A common driveway, approved under 2.4.1.2 may serve:

- three or four lots if at least one is a backland lot.
- three, four or five lots if two or more are backland lots.

Backland Lots are as defined in and permitted by Special Permit issued under the Town of Bolton's Zoning Bylaw 2.3.5.5 Backland Zoning.

A lot served is any lot crossed by, whether or not any building or any dwelling on the lot, is actually accessed and/or served by this Common Driveway, or lot on which any building or any dwelling is accessed and/or served by this Common Driveway. All such lots must be included in the list and number of lots served.

(b) Each lot served by the common driveway must have permanent access to the Common Driveway pursuant to an Easement Agreement acceptable to the Planning Board and the Deed to each lot served on a Common Driveway must reference this Easement Agreement. The Easement Agreement is to be recorded along with the Special Permit with the Worcester Registry of Deeds or with the Worcester County Land Court.

(c) Any deeds of ownership of lots served by a common driveway shall require that the owners of said lots must be members of a maintenance association, whose purpose is to provide for maintenance of the common driveway, which shall include, but not be limited to, snow plowing and maintaining design specifications. This maintenance association must be created by a Maintenance Association Agreement acceptable to the Planning Board and the Deed to each lot served on a Common Driveway must reference this Maintenance Association Agreement. This Maintenance Association Agreement is to be recorded along with the Special Permit with the Worcester Registry of Deeds or with the Worcester County Land Court.

(d) The common driveway is defined as extending from the approved or accepted right-ofway to which it is attached, to the point it serves only one lot, the so-called terminus. A Common Driveway may have more than one terminus. The common driveway shall be connected to an approved or accepted right-of-way at one, and only one, point. The entire common driveway must lie within the lots served, and in the Town of Bolton.

(e) The common driveway must meet the design criteria of this bylaw, and any additional design criteria established by the Planning Board in regulations duly voted by said Board according to law. The design criteria of this bylaw are:

- Twelve (l2) feet minimum width of wear surface.
- A minimum of eight (8) inches of gravel wear surface.
- The first forty (40) feet of the common driveway from an approved or accepted right-ofway must have a slope of four (4%) percent or less.
- Suitable drainage appurtenances to prevent excessive erosion. These Drainage appurtenances must further ensure to the satisfaction of the Planning Board that all proposed activity and all development, including the construction of dwellings, lawns and other impervious areas for all lots to be served by this Common Driveway, in no manner contribute to additional drainage onto any abutting property or onto any accepted or approved right of way.
- A maximum slope of twelve (12%) percent.
- The centerline intersection with an approved or accepted right-of-way must be sixty (60) degrees or more.
- A turnaround located near each terminus, which location must be acceptable to the Planning Board, of at least forty (40) feet width and thirty (30) feet depth.
- An intercept width with the approved or accepted right-of-way of at least fifty (50) feet.

- A staging area of at least forty (40) feet in length and a minimum of twenty (20) feet in width at the streetline, tapering to a minimum of twelve (12) feet in width at forty (40) feet from the streetline.
- Passing turnouts must be constructed which provide a total width of at least eighteen (18) feet along a distance of at least twenty five (25) feet, spaced no more than three hundred (300) feet between turnouts or at a lesser interval where in the Planning Board's opinion a lesser distance is warranted for safety considerations.
- The length must be such that the distance along the Common Driveway centerline to each building or dwelling served by the Common Driveway will not exceed 1800 feet from the street sideline, and that the length along any of its individual driveways measured from the centerline of the Common Driveway to any building or dwelling served by the Common Driveway shall not exceed 800 feet.
- Signs to direct emergency access must be installed at the streetline and at each driveway intersection with the Comon Driveway.
- The centerline of the Common Driveway cannot be located closer than thirty five (35) feet to the centerline of any approved or constructed single dwelling driveway or Shared Driveway.

When deciding whether or not to grant a special permit to create a common driveway, the Planning Board should consider:

- the safety of the common driveway as designed, for normal use.
- the safety of the intersection with the town way.
- the adequacy of the legal agreements for maintenance and access.
- the adequacy of the common driveway to provide access to vehicles carrying materials which are potentially hazardous if spilled, such as home heating oil.
- the environmental impact on wetlands and water resource areas.

(f) Granting of a Special Permit under this bylaw does not constitute a waiver of any other applicable Bylaw or Statute.

The Planning Board shall issue regulations to go with this bylaw, including charging a filing fee adequate to cover both legal review and construction inspection.

The driveway shall be sufficiently constructed for access before issuance of any building permit.

2.4.1.3 Parking

(a) A parking area, (not including access ways) of at least one square foot for each one square foot of business and commercial building area and one square foot for each two square feet of industrial building floor area, and three square feet for each one square foot of retail building floor area shall be provided on each lot of land used as an industrial, commercial or business area.

The parking area is defined to include the parking spaces and any aisles between the spaces, but not any driveways leading to the parking area.

The parking area shall not come within thirty feet (30') of the edge of the street paving.

(b) Parking areas and driveways on business, industrial or commercial sites shall be improved to at least the equivalent of eight inches (8") gravel over sub-grade and shall be pitched to drain.

2.4.2 Sign Regulations.

After January 25, 1971 no person shall place or cause to be placed, posters, handbills, placards, advertising matter or signs advertising or indicating the location of a personal, industrial or commercial enterprise or products anywhere in the Town except by written permission of the Board of Selectmen and:

- 2.4.2.1 No sign shall exceed thirty six (36) square feet.
- 2.4.2.2 No sign shall be oscillating, flashing, operated with moving parts or display the illusion of motion.
- 2.4.2.3 No sign shall be closer than twenty (20) feet from the road pavement.
- 2.4.2.4 No sign including standards shall be more than eight (8) feet in height.

2.4.2.5 Signs in Residential Zones.

(a) Residences may have by permit, one sign, non-illuminated and of area not greater than two square feet for the purpose of announcement of professional or home occupation or for the announcement of the occupants of the dwelling. The sign shall be limited to one back-ground color which shall be white, natural wood or the color of the principal structure or its trim and any color for all lettering and other designs. The lettering on the sign shall not exceed three inches in height. The sign must not be closer than twenty feet from the side lot line. The sign must not be closer than ten feet from the road pavement.

(b) Pre-existing non-conforming uses in residential zones desiring a new sign not conforming under section (a) may apply under the general Bylaw.

- 2.4.2.6 Temporary signs, pertaining only to the lease or sale of land or building on which the sign is located, will be allowed in all zones subject to the following restrictions:
 - (a) The sign shall not exceed six square feet in area.
 - (b) The sign shall not be illuminated.
 - (c) The addition of the word "sold" is not allowed.

Signs which do not conform to this section must apply under the General Sign Bylaws (Section 2.4.2.).

- 2.4.2.7 If literal enforcement results in substantial hardship and if no harm to the public good is evident, waivers may be granted by the Board of Selectmen.
- 2.5 SPECIAL REGULATIONS
- 2.5.1 Trailers and Mobile Homes
- 2.5.1.1 Bylaw Applicability

All trailer coaches or mobile homes, used for dwelling, commercial, business or industrial purposes, parked in any area of the Town for more than ninety (90) days shall conform to all Bylaws and regulations applying to permanent dwellings or buildings except in emergencies as provided by the fourth paragraph of Section 3, Chapter 40A of the General Laws.

2.5.1.2 Time Restriction

The use of trailers or mobile homes for dwelling, commercial, business or industrial purposes in the Town for more than a period of one year shall be prohibited.

2.5.2 Accessory Uses

2.5.2.1 Home Occupations

In all Districts professional office or customary home occupation uses are permitted provided that:

(a) The profession or home occupation is conducted by a resident of the premises.

(b) The use is clearly incidental to and secondary to the use of the premises for residential purposes and the external character of the premises is that of a one-family residence.

(c) Not more than two (2) persons other than the residents of the dwelling and not more than a total of four (4) are employed at any one time on the premises in the home occupation.

(d) No noise, vibration, dust, heat, odors, glare, traffic congestion, unsightliness or other nuisance results which is discernible from other properties, or which is detrimental to the environment.

2.5.2.2 Swimming Pools

No person shall erect or construct a permanent swimming pool in the Town of Bolton without a building permit. Outdoor swimming pools shall be protected by a fence, wall, building, enclosure or solid wall of durable material of which the pool itself may be constructed, or any combination thereof. Artificial barriers shall be constructed at least four (4) feet in height so as to be impenetrable by toddlers. These barriers should either completely encircle the pool or the yard in which the pool is located. Any entrances must be equipped with a self-closing and positive self-latching closure mechanism at a height of 42 inches or more, and provided with hardware for permanent locking.

2.5.2.3 Accessory Scientific Uses

Scientific research or development may be allowed on Special Permit as an accessory use in any district, as provided in paragraph 9, Section 9, Chapter 40 A, General Laws.

2.5.2.4 Accessory Apartments

Construction of an accessory apartment is allowed, either in, or attached to, a new or existing dwelling subject to the following requirements:

(a) The residence must be owner-occupied.

(b) Size of the accessory apartment is to be limited to no larger than one-third of the floor space of living area of the residence.

(c) The outside appearance of the premises shall remain that of a single family residence.

(d) All applicable federal, state and local building and health codes must be satisfied including all bylaws of the Town of Bolton.

(e) The accessory apartment shall have its own separate entrances from the outside.

(f) The accessory apartment shall have its own complete kitchen and complete bath and toilet facilities.

(g) There will be only one accessory apartment per residence.

(h) A certified drawing showing the above shall be filed with the Board of Selectmen either for new construction or the creation of an accessory apartment in an existing dwelling.

2.5.2.5 Agricultural /Business Use

The Town of Bolton finds that in order to protect and preserve agricultural, horticultural, floricultural, or viticultural lands, to preserve natural resources and maintain land in active agricultural, horticultural, floricultural, or viticultural use, it is necessary to allow the owners of said lands to conduct an accessory business to supplement income from said uses. Agricultural/Business Use may be allowed by Special Permit issued by the Board of Selectmen under Section 2.2.3 as an accessory use in any district.

a) Allowable Agricultural/Business Uses. Allowable accessory uses are the following:
1) retail sales of farm products, crafts and similar retail products, as well as newspapers or magazines, including serving and consumption of baked goods and other prepared foods on the premises during hours of operation as defined in the Special Permit;
2) large and small animal veterinary services;

3) hosting or staging of revenue-generating events, tours, weddings, and functions which are appropriate in scale to the premises and surrounding residential area, including the preparation and serving of food and beverages for such events

4) Wireless Communication Facility(s), as defined in 2.5.7 Wireless Communication Bylaw, and any accessory structure(s) and or building(s) replaced, constructed, installed and or maintained pursuant to Special Permit(s) issued under 2.5.2.5 Agricultural/Business Use and 2.5.7 Wireless Communication Bylaw.

b) Prohibited Uses. Specific uses which are not allowed on residentially-zoned agricultural land are the following:

1) Gas stations or retail or wholesale fuel storage or delivery;

- 2) Retail stores such as drug, department, pastry, hardware, and clothing stores; and
- 3) Manufacture of non-agricultural products.

All other uses must meet the requirements of Section A. above.

c) Eligibility. An owner may apply for an Agricultural/Business Use Special Permit if the owner has not fewer than 20 acres on one or more contiguous parcels or 75 contiguous acres as specifically required by 2.5.2.5 e) Wireless Communication Accessory/Business Use within the Town of Bolton which is:

- 1) being actively farmed, or which has a specific farming plan submitted with the application, and
- 2) which has a state, town, or privately held perpetual agricultural preservation restriction or conservation restriction or an application for such a restriction pending before the appropriate approving authority for the restriction. The restriction must apply to at least 20 acres, but the accessory use may be on un-restricted land contiguous to the agricultural land. Any special permit granted regarding land which is the subject of a pending restriction will be issued conditionally upon approval of the restriction.

d) Criteria. The proposed agricultural/business use must be clearly accessory to the principal use of the premises. The special permit shall be granted only upon the determination that:

- the use meets the definition of Section A above and is not one of the excluded uses in Section B. above; and
- 2) the use is reasonably related to the primary use and serves to promote the primary use and its products beyond merely bringing in additional revenue; and
- 3) the use is not detrimental to the primary use of the land; and
- 4) total annual projected sales from the accessory use when fully operational do not exceed total sales derived from agriculture, horticulture, floriculture, or viticulture; and
- 5) facilities and structures constructed for the accessory use are compatible with other agricultural facilities on the property in function, visually, and in scale such that the total footprint of all accessory facilities does not exceed 50% of the footprint of all agricultural facilities on the property; and
- 6) the accessory use is:
- (a) appropriate to the site;
- (b) meets all of the requirements of Section 2.2.3.2 of the bylaws;
- (c) does not substantially adversely affect the surrounding residential properties as compared with the primary use as to sound, light, odor, noise, and other disturbances;
- (d) does not pose a threat to traffic safety; and
- (e) does not adversely affect the Town's ability to provide municipal and public safety services to the premises by way of existing roads and with the use of existing town equipment.

e) Wireless Communication Agricultural/Business Use. An Agricultural/Business Use Special Permit to construct a Wireless Communication Facility(s) pursuant to this bylaw may be issued if in addition to the above, the owner and the parcel(s) on which the Wireless Communication Facility(s) is (are) located have complied with the following:

1) a Special Permit issued under 2.5.7 Wireless Communication Bylaw has previously or is contemporaneously granted; and

2) the property has (i) an Agricultural Preservation Restriction with the Commonwealth of Massachusetts in perpetuity pursuant to Massachusetts General Laws, or (ii) an Agricultural Development Restriction with the Town of Bolton, or (iii) an application for such a restriction pending before the appropriate approving authority.

An Agricultural Development Restriction is a restriction granted to the Town of Bolton that is administered by the Board of Selectmen and recorded with the Worcester Registry of Deeds or with the Land Court that prohibits the subdivision, development or other non-agricultural use of any part of the restricted property. The Agricultural Development Restriction must contain wording that specifies that the Board of Selectmen will not release any of the restricted property until they have determined that the property has been restored to substantially the same condition as it was in prior to the construction of the Wireless Communication Facility. The Agricultural Development Restriction must further convey to the Town of Bolton in perpetuity a right of first refusal to acquire the restricted property.

Any Special Permit granted which is the subject of a pending restriction will be issued conditionally upon approval of the restriction. The executed restriction must be recorded with the Worcester Registry of Deeds; and

- 3) for each Wireless Communication Facility, the Agricultural Preservation Restriction or the Agricultural Development Restriction must apply to contiguous parcel(s) of at least seventy-five (75) contiguous acres within the Town of Bolton and 2.5.2.5 c) 2) does not apply. More than one Wireless Communications Facility or Tower may be situated on a contiguous parcel(s). For each additional Tower after the first Tower on a contiguous parcel(s), the contiguous property under either an Agricultural Preservation Restriction or the Agricultural Development Restriction must be increased by a multiple equal to at least the number of additional Towers times a minimum of seventy-five (75) contiguous acres within the Town of Bolton; and
- 4) no portion of the property included in the area restricted pursuant to 2.5.2.5 e) 2 can be included or considered as part of any other Agricultural Preservation Restriction for any other allowable Agricultural/Business Use application or Special Permit under 2.5.2.5.
- 5) all of the Wireless Communication Facility(s) must be located wholly within the contiguous restricted property; and
- 6) the restricted property must have at least 200 feet of continuous frontage on an approved or accepted right of way within the Town of Bolton; and
- 7) any Wireless Communication Facility must be set back from any:

i) property line, other than a property line immediately bordering Rt. 495, by not less than 600 feet. From a property line bordering Rt. 495 the setback shall be at least one (1) time the height of the Wireless Communication Facility, including any appurtenant equipment or Communication Device(s) attached thereto; and

ii) the centerline of an approved or accepted right of way, other than Rt. 495, by not less than 800 feet. Adjacent to Rt. 495 the property line setback will apply; and

iii) the Wireless Communication Bylaw 2.5.7.6 Required Findings For A Special Permit 8(i) and 8(ii) do not apply; and

iv) At the time of the Special Permit Application, from any residence which has been built or for which a building permit has been granted or from the site of any residence shown on a plan of land approved by the Town of Bolton or under consideration for approval by the Town of Bolton by not less than 1,000 feet.

8) the requirements of 2.5.2.5 d (b) 2) and 2.5.2.5 d (c) shall not apply.

Sufficient grounds for the immediate revocation of the Agricultural/Business Use Special Permit for a Wireless Communication Agricultural/Business Use shall include the discontinuation of the agricultural use of the restricted property.

Administration. Such permit may also impose conditions, safeguards and limitations on time and use. Changes to the accessory use and/or increases to and/or size of agricultural facilities

which result in increases in the number or size of accessory facilities require submission of a new Special Permit application to the Special Permit granting authority.

f). Enforcement. The Special Permit granting authority shall monitor conduct of agricultural/ business accessory uses for compliance with the terms of the special permit and any conditions attached thereto. The Special Permit granting authority shall impose as a condition that the permit holder must provide, if requested, annual sales information for the accessory agricultural/business and primary agricultural use of land. If the Special Permit granting authority determines that the permit conditions have been violated, the permit may be revoked following notice and public hearing.

- 2.5.3 Flood Plain District
- 2.5.3.1 The Flood Plain District is herein established as an overlay district. The underlying permitted uses are allowed provided that they meet the following additional requirements as well as those of the Massachusetts State Building Code dealing with construction in flood plains. The Flood Plain District includes all special flood hazard areas designated as Zone A, A1-30 on the Town of Bolton Flood Insurance Rate Maps (FIRM), and the Flood Boundary and Floodway Maps, dated June 18, 1980, on file with the Town Clerk, Planning Board and Building Inspector. These maps as well as the accompanying Bolton Flood Insurance Study are incorporated herein by reference.

2.5.3.2 Development Regulations

The following requirements apply in the Flood Plain District:

(a) Within Zone A, where the base flood elevation is not provided on the FIRM, the applicant shall obtain any existing base flood elevation data and it shall be reviewed by the Building Inspector for its reasonable utilization toward meeting the elevation of flood proofing requirements, as appropriate, of the State Building Code.

(b) In the floodway, designated on the Flood Boundary and Floodway Map, the following provisions shall apply:

1. All encroachments, including fill, new construction, substantial improvements to existing structures, and other development are prohibited unless certification by a Registered Professional Engineer is provided by the applicant demonstrating that such encroachment shall not result in any in-crease in flood levels during the occurrence of the 100-year flood.

2. Any encroachment meeting the above standard shall comply with the flood plain requirements of the State Building Code.

2.5.3.3 Applicability

Nothing contained in this section shall limit the authority of the Board of Health with respect to premises in the Flood Plain District or affect the applicability of any other Bylaw to any building activity in the Flood Plain District.

- 2.5.4 Deleted Annual Town Meeting May 1, 2006 (Water Resource Protection District)
- 2.5.5 Business, Commercial and Industrial Regulations

2.5.5.1 General

Any structure or outdoor storage for business, commercial or industrial use shall comply with the following requirements:

2.5.5.2 Activity Regulations

No such use shall:

- (a) Store or use volatile poisonous liquids in excess of 20,000 gallons.
- (b) Operate slaughter houses commercially.
- (c) Store or produce explosives.
- (d) Wreck or store junk or wreckage of any kind.
- (e) Store or produce fertilizers.
- (f) Store or produce any product which might be noxious, pungent, or create ugly refuse.
- (g) Produce objectionable noise, dust or dirt.
- (h) Produce a product or by-product emitting harmful radiation.
- (i) Create any form of environmental pollution.
- (j) Involve the sale of new or used motor vehicles.
- 2.5.5.3 Greenspace Regulations

(a) Landscaping: All areas not covered by structures or used for access or parking shall be protected by grass, trees, shrubbery, or equivalent cover.

(b) Buffer: Structure or outdoor storage shall be separated from any lot residentially used or zoned by a buffer zone either naturally wooded for fifty feet (50') in width, or if approved by the Board of Selectmen, by a narrower buffer providing equivalent visual screening through densely planted evergreen trees large enough to reach five feet (5') in height in three years and of a type which will continue to grow in height.

2.5.5.4 Firefighting Water

An accessible tank, reservoir, or water hole containing at least one gallon of water for each square foot of total floor area of buildings of combustible material and one gallon of water for each two square feet of total floor area of buildings of noncombustible material shall be provided and maintained within 1,000 feet of each industrial or commercial building unless said building is located within 1,000 feet of an existing equivalent body of water.

2.5.5.5 Special Permits For Zoning Use

For those uses identified in Section 2.3.4 of the Bolton Zoning By-laws as requiring a special permit from the Planning Board, the following factors shall be considered by the Planning Board in granting, denying, or considering renewal of any special permit:

(a) the existence of safe vehicle access to and from the right-of-way;

(b) the existence of safe pedestrian access to and from the site;

(c) the adequacy of provisions to reduce or eliminate undesirable visual, noise, odors or similar impacts upon adjoining properties and the public;

(d) the adequacy of provisions for controlling and cleaning up on-site and off-site litter and debris;

(e) the existence of water and waste management plans for the premises which will minimize any adverse impact on natural and community resources;

(f) the adequacy of proper controls for containment of exhaust fumes or other emissions from the premises;

(g) the degree to which the project will recycle waste materials, will utilize recycled materials, and will utilize water conserving and energy efficient appliances;

(h) the degree to which the project design is in harmony with the neighborhood and the environment; and

(i) whether in all other respects the proposed project will be in harmony with the general purpose and intent of the Bolton Zoning by-law and not detrimental to the neighborhood or the Town.

Any special permit granted under this section may contain such conditions, restrictions or requirements as the Planning Board deems appropriate to accomplish the purposes and intent of this Zoning By-law and to assure that the proposed use will satisfy the criteria set forth above.

In addition to the special permit required under this section, the applicant shall also meet the requirements of the Site Plan Approval Process before the Board of Selectmen as set forth in Section 2.5.5.6 of the Bolton Zoning Bylaws.

2.5.5.6 Site Plan Approval Process.

To assist in assuring compliance with the Zoning Bylaw and other provisions of the law, any use subject to Site Plan approval shall be established or expanded in ground area and any building for such use shall be erected or externally altered only in conformity with a Site Plan approved by the Board of Selectmen with the advice of the Planning Board. Site Plan approval will be required of all uses in Limited Business, Limited Recreation Business, Business, Commercial and Industrial Zones, exempting agricultural, horticultural, floricultural uses.

All Site Plans (four copies) shall be submitted directly to the Board of Selectmen. When the Board of Selectmen receives a Site Plan for its approval, it shall deliver a copy to the Planning

Board for advice. The Board of Selectmen shall act on the Site Plan only after the Planning Board has submitted its advice, or has allowed thirty (30) days to pass without acting. Where a Site Plan is required, the Inspector of Buildings shall issue a building permit only on receipt of an approved Site Plan and a Certificate of Occupancy shall not be issued until completion of all improvements and compliance with the approved Site Plan. The Board of Selectmen and the Inspector of Buildings shall maintain permanent files of Approved Site Plans. The Board of Selectmen shall either approve or note deficiencies within fourteen days of receipt of Planning Board review. Where appropriate, the Board of Selectmen, with advice of the Planning Board may waive the requirements for the Site Plan application.

The Site Plan shall show information as needed to determine compliance with the Zoning Bylaw and shall be prepared by a Registered Professional Engineer, Land Surveyor, Architect, and/or Landscape Architect.

In reviewing the Site Plan, the Planning Board and the Board of Selectmen shall consider the following matters:

(a) Protection of adjoining premises against detrimental uses by provision for surface water drainage, sound and sight buffers, generation of electrical machine interference, preservation of views, light and air;

(b) Convenience and safety of vehicular and pedestrian movement within the site, the location of driveway openings in relation to traffic or to streets serving the site, and when necessary, compliance with other regulations for the handicapped, minors and the elderly, including the location of pedestrian access adjacent to the site;

(c) Adequacy and the arrangement of parking and loading spaces in relation to the proposed uses of the premises and the abutting property;

(d) Adequacy of the methods of disposal of refuse and other wastes resulting from the uses permitted on the site;

(e) Relationship of structures and open spaces to the natural and planned landscape, existing buildings and other community assets in the area and compliance with other requirements of the Zoning Bylaws;

(f) Demands on the Town's resources including effect on the Town's water supply, fire protection, police enforcement and highway department;

(g) Location, size, lighting and appearance of all site signage, traffic and parking lights, fencing, petroleum and chemical storage facilities, fire and life safety equipment.

(h) And may impose such appropriate conditions, limitations, and safeguards as will insure compliance with the terms of the approval.

(i) Regulations and a filing fee may be adopted by the Board of Selectmen.

Where a Site Plan as submitted requires, revision and deficiencies shall be indicated to the applicant clearly in writing, but Site Plan approval shall not be granted until all required changes have been incorporated on a resubmitted plan; any resubmitted plan shall be subject to the procedure outlined in paragraph 2 above. Site Plan approval shall expire if any work thereunder is not begun within twelve (12) months after issuance of the building permit and not diligently carried through to completion.

2.5.6 Scenic Roads

Acting under Section 15C, Chapter 40 of the General Laws, the following roads are designated as Scenic Roads:

Ballville	Old Bay
Bear Hill	Old Harvard
Berlin	Old Sugar
Bolton St.Rd.No.1	Quaker
Burnam	Randall
Century Mill	Sampson
Corn	Sargent
East End	Sawyer
Flanagan	South Bolton
Forbush Mill	Spectacle
Frye	Sugar
Golden Run	Teele
Green	Town Farm
Jordan	Vaughn Hill
Lewis	Warner
Lively	Wattaquadock (From Old Bay to Main)
Long Hill	West Berlin
Manor	Wheeler
Meadow	Whitcomb
Moore	Wilder
Nourse	Willow

Designation as a Scenic Road allows the Town to preserve the qualities and character of the town ways. Any repair, maintenance, reconstruction, or paving work done with respect thereto, of a Scenic Road shall not involve or include the cutting or removal of trees, or the tearing down or destruction of stone walls, or portions thereof, except with the prior written consent of the Planning Board after a Public Hearing duly advertised.

In granting or refusing such consent the Planning Board shall consider, among other things, the public safety, scenic views, preservation of historic and regional characteristics, and preservation and enhancement of natural and aesthetic qualities of the environment.

The Planning Board may adopt reasonable further standards relative to Scenic Roads not inconsistent with this Bylaw and the General Laws.

2.5.7 Wireless Communication Bylaw

2.5.7.1 Purpose

The purpose of the Wireless Communication Bylaw is to establish predictable and balanced regulations for the siting of wireless communication equipment in order to accommodate the growth of wireless communication systems within the Town of Bolton while protecting the public against adverse impacts upon the Town's aesthetic resources and the public welfare and to:

- 1. Promote the public health, safety, general welfare, quality of life and rural character of the Town of Bolton;
- 2. Guide sound development;
- 3. Conserve the value of lands, natural resources, residences and buildings;
- 4. Encourage the most appropriate use of the land;
- 5. Minimize the adverse aesthetic and visual impact of wireless communication facilities;
- 6. Minimize the number of wireless communication facility sites;
- 7. Encourage co-location of licensed wireless communication carriers;
- 8. Ensure that wireless communication facilities are sited, designed and screened in a manner that is sensitive to the surrounding neighborhood and the Town of Bolton;
- 9. Avoid damage to and limit the impact upon abutting properties.

2.5.7.2 Definitions

- Communication Device Any antennae, dish or panel or similar equipment mounted out of doors on a tower, building or structure used by a licensed commercial telecommunications carrier(s) to provide telecommunication(s) services. The term Communication Device does not include a tower. High gain point to point antenna are not permissible.
- Tower -. Any equipment mounting structure that is used primarily to support any reception equipment, transmission equipment or Communication Device that measures twelve (12) feet or more in its longest vertical dimension. The definition of tower shall include, but not be limited to, monopole and lattice structures.
- Wireless Communication Facility Any and all construction, installation, expansion, extension or use of any Communication Device(s), Tower(s), materials, equipment, storage structures, accessory buildings, dishes and antennas used by a licensed commercial telecommunication carrier(s), the Town of Bolton's municipal public safety services and/or the Town of Bolton's public schools to provide telecommunication services including personal wireless communication.

2.5.7.3 Applicability and Special Permit

No Wireless Communication Facility shall be erected, installed, modified, replaced or maintained except upon the grant of a Special Permit in compliance with the provisions of this Wireless Communication Bylaw. The provisions of the Wireless Communication Bylaw will apply to all Wireless Communication Facilities whether as a principal use or an accessory use and to any and all modification, extensions, and additions to, or replacements of existing Wireless Communication Facilities.

All modifications, extensions and additions to, or replacements of a Wireless Communication Facility shall be subject to a modification of the Wireless Communication Bylaw Special Permit following the same requirements as required for an original application.

A Wireless Communication Bylaw Special Permit may be granted only for:

A. Any Wireless Communication Facility to be constructed, installed, replaced, maintained and/or used in the business, limited business, commercial and industrial zoning districts as specifically provided for hereafter, in compliance with the provisions of this Wireless Communication Bylaw and upon the grant of a Special Permit.

B. Any Wireless Communication Facility to be constructed, installed, replaced, maintained and/or used in the residential zoning district as specifically provided for hereafter in conformance with 2.5.2.5 Agricultural/Business Use and in compliance with the provisions of this Wireless Communication Bylaw and upon the grant of a Special Permit.

No Wireless Communication Facility shall be erected, installed, modified, replaced, maintained and/or used in a Wetland Resource Area as defined in the Wetlands Bylaw 1.18.2.

2.5.7.4. Jurisdiction

The Board of Selectmen is authorized to grant and modify Special Permit(s) pursuant to Massachusetts General Laws Chapter 40A, for Wireless Communication Facility(s) within the Town of Bolton. The Board of Selectmen are further authorized to grant or modify Special Permit(s) for the construction, installation, replacement, maintenance and/or use of Wireless Communication Device(s) and Wireless Communication Facility(s) on already existing buildings or structures in the residential, business, limited business, commercial or industrial zoning districts.

The Wireless Communication Bylaw Special Permit will establish (i) the intensity of use, including the number, type and location of Wireless Communication Device(s), (ii) periodic monitoring and reporting and (iii) other provisions as determined by the Board of Selectman.

A Wireless Communication Facility shall not be required to be sited on its own lot as required by 2.3.5.3 One Building Per Lot.

To facilitate the administering of this Wireless Communication Bylaw, the Board of Selectmen may establish and amend: (i) rules and regulations, (ii) application fees, requirements and procedures, (iii) engineering, consulting, review, and periodic monitoring fees, and (iv) reasonable Special Permit license fees.

2.5.7.5 Expiration

Any Wireless Communication Special Permit granted or renewed pursuant to this Bylaw shall expire five (5) years after the date of the decision of the Board of Selectmen or the determination of an appeal as referred to in Massachusetts General Laws, Chapter 40A, Section 17.

An application to renew or extend a Special Permit granted pursuant to this bylaw will be considered and administered as a new Special Permit application.

2.5.7.6 Required Findings For A Special Permit

The Board of Selectmen may grant or modify a Wireless Communication Bylaw Special Permit only if they find:

- 1. That there is no existing or approved Wireless Communication Facility(s), Tower(s), or other structure(s) reasonably available that could accommodate the Wireless Communication Facility or could reasonably provide the service(s) or a reasonable alternative service.
- 2. That the applicant demonstrated to the satisfaction of the Board of Selectmen that the siting and proposed location of the Wireless Communication Facility is critical and cannot be reasonably accommodated by co-locating on existing Wireless Communication Facilities.
- 3. That the applicant and all tenants hold at the time of application all necessary Federal, State and FCC telecommunications licenses required to operate the Wireless Communication Facility.
- 4. That the applicant has certified to the Board of Selectmen that they are in compliance with all applicable Federal, State and local laws and regulations including any amendment(s) thereto.
- 5. That the size and height of the Wireless Communication Facility is the minimum necessary to accommodate all users thereon and that any Tower, including all appurtenant equipment, Wireless Communication Device(s) and or Communication Device(s) shall:
 - i) Not exceed seventy-five (75) feet above the average grade of the existing terrain at the tower's base, unless the applicant demonstrates to the satisfaction of the Board of Selectmen that a taller structure is required.
 - ii) In no event exceed a height of one hundred and fifty (150) feet above the average grade of the existing terrain at the tower's base; and
 - iii) In no event be of such a height or location as to require aviation warning lighting.
- 6. That the Wireless Communication Facility has been designed in all respects to accommodate multiple licensed telecommunication carriers and that the applicant has agreed to and will permit other licensed telecommunications carriers to co-locate on the facility at commercially reasonable terms.
- 7. That the Wireless Communication Facility will not have a material adverse impact upon any Bolton viewshed as determined solely by the Board of Selectmen.
- 8. That any Wireless Communication Facility in the business, limited business, commercial or industrial zoning districts is set back from:
 - Any property line, other than a property line immediately bordering Rt. 495, the greater of 400 feet or a distance equal to two (2) times the height of the Wireless Communication Facility, including any appurtenant equipment, device(s) or Wireless Communication Device(s) attached thereto. From a property line bordering Rt. 495 the setback shall be at least one (1) time the height of the Wireless Communication Facility, including any appurtenant equipment, device(s) attached thereto. Setting Rt. 495 the setback shall be at least one (1) time the height of the Wireless Communication Facility, including any appurtenant equipment, device(s) or Wireless Communication Facility, including any appurtenant equipment, device(s) or Wireless Communication Device(s) attached thereto; and
 - ii) The centerline of an approved or accepted right of way, other than RT. 495, by not less than 500 feet; and
 - iii) Any residence which has been built or for which a building permit has been granted at the time of the application, or from the site of any

residence shown on a plan of land approved by the Town of Bolton or under consideration for approval by the Town of Bolton at the date of application, by not less than 1,000 feet.

- 9. That any Tower has been designed, using the best available technology to blend in to the surrounding environment through the use of color, camouflaging techniques, or other architectural treatments. Monopole construction is preferred.
- 10. That the Wireless Communication Facility includes a provision(s) for the reasonable co-location at no cost to the Town of Bolton of communication systems to support the Town's emergency and safety services and its municipal public schools.
- 11. That the Wireless Communication Facility complies with the following minimum design requirements:
 - a) that the Wireless Communication Facility has been so sited so as to make use of natural vegetative screening and that an adequate replacement of vegetation providing a noise buffer to neighboring properties is provided for, and
 - b) that to the greatest extent practical the clearing of existing vegetation and the impact on the site's natural resources and topography is minimized, and
 - c) that any clearing of existing on-site vegetation will preserve such vegetation to the maximum extent practicable and that any disturbed areas will be restored to the maximum extent practicable, and
 - d) that access to any Wireless Communication Facility has been engineered and will be built to ensure that the Town of Bolton's emergency services can respond safely to the site, and
 - e) that emergency access to the site is available at all times to the Town of Bolton's emergency services, and
 - f) that any emergency backup generators will be installed such that they only cycle periodically during non-holiday weekdays, between 8:00AM and 5:00PM, and
 - g) that any wireless communication facility or tower is fenced so as to control access to the facility and that any fencing is designed so as to be as unobtrusive as possible, and
 - h) that only signage acceptable to the Board of Selectman will be visibly posted. At a minimum an announcement sign, a no trespassing sign and a sign giving a phone number where the owner or operator can be reached on a twenty-four hour basis, and
 - that any accessory buildings and structures (i) do not exceed one story in height, (ii) contain no more than 300 square feet in floor area for each user, (iii) is located within the fenced in area, and (iv) is compatible in appearance.

2.5.7.7 Non Use

The operator must provide immediate acceptable written notice to the Board of Selectmen in the event of any change of ownership in any wireless communication facility, tower, communication device, accessory building, or of the underlying property. The operator or the property owner must provide immediate notice to the Board of Selectmen if the use of any wireless communication facility, tower, communication device or accessory structure is discontinued. Any unused wireless communication facility, tower, communication

device and/or accessory structure, shall be removed from the Town of Bolton within one (1) year of cessation of use and the property shall be restored to substantially the same condition as it was in prior to the alteration.

To secure compliance with this provision, the Board of Selectmen may require that the applicant post adequate and acceptable surety as determined solely by the Board of Selectmen.

All unused wireless communication facility(s), tower(s), communication device(s), accessory structure(s) and or any parts thereof, which have not been used for two years and which have not been removed from the Town of Bolton, may be dismantled, removed and the site restored to substantially the same condition as it was in prior to the alteration by the Town of Bolton. All costs incurred by the Town of Bolton directly related to and incidental to any such dismantling, removal and restoration shall be payable by the owner of the property.

As conditions to any Special Permit granted under this Wireless Communication Bylaw the property owner shall be required (i) to enter into an agreement with the Town of Bolton whereby the property owner is liable for all costs set forth in the preceding sentence and agrees to the conditional lien on the property as set forth below; (ii) to execute a document creating a lien on the property which by its terms shall become effective in the event that the Wireless Communication Facility has not been used for two years and has not been removed and the site restored within thirty days of written notice by certified mail to the owner; and (iii) to execute a Notice of Contract regarding the real estate. Both the lien document and the Notice of Contract shall be in a form suitable for recording at the Registry of Deeds. The Notice of Contract shall be recorded at the time of recording the Special Permit; the lien document shall be recordable upon the attachment of an affidavit signed by the Chair of the Board of Selectmen that the Wireless Communication Facility was unused for two years and has not been removed by the owner and setting forth as the amount of the lien a sum equal to all costs incurred by the Town of Bolton to dismantle and remove the facility and restore the property.

2.5.7.8 Compliance

The failure of the applicant, owner, owner of the property, and or any licensed telecommunications carrier tenant to comply with the Bylaws of the Town of Bolton or with any section of the Wireless Communication Bylaw Special Permit shall be sufficient grounds for the immediate revocation or non-renewal of the Special Permit.

2.5.7.9 Exemptions

The following are exempted from this Wireless Communication Bylaw:

1. Amateur radio tower or communications device(s). An amateur radio tower or communications device(s) defined as a Tower or Communication Device(s) used solely in accordance with the terms of an amateur radio license(s) issued by the Federal Communications Commission. In order to qualify for this exemption, the Tower and any Communication Device(s):

- i) must not be used or licensed for any commercial purposes; and
- ii) must be immediately dismantled if the amateur radio license is revoked or not renewed by the FCC.

2. Wireless Communication Facility(s), Tower(s) or Communication Device(s) erected and maintained by the Town of Bolton and/or the Town of Bolton's public schools solely for the Town of Bolton's municipal emergency and safety communication purposes. In the event that any Wireless Communication Facility(s), Tower(s) or Communication Device(s) exempted by this section is no longer used, or any part thereof is no longer used, the unused portion must be removed within one (1) year as provided for and in compliance with Section 2.5.7.7 Non Use.

Lawfully preexisting Wireless Communication Facility(s), Tower(s) or Communication Device(s) and any accessory structure(s) may not be modified, replaced, extended or added to except in accordance with 2.1.3.3. Pre-existing Nonconforming Uses and Structures and, if applicable, this Wireless Communication Bylaw. For any pre-existing nonconforming Wireless Communication Facility(s), Tower(s) or

Communication Device(s) the provisions in 2.1.3.3. a) Extension or Alteration that considers any increase of not over 100% from the original floor area of the building at the time of the adoption of the Zoning Bylaw on April 13, 1972, or not over 50% of the ground area in use at that time as not substantially more detrimental to the neighborhood will not apply. Any increase will need to conform with the requirements of 2.5.7 Wireless Communication Bylaw.

2.5.8 RATE OF DEVELOPMENT

2.5.8.1 Purpose and Intent

The purpose of this bylaw is to ensure orderly growth and development consistent with the Town's historic rate of residential growth for a limited period of time to permit the Town of Bolton to comprehensively address issues created by residential and commercial growth through the creation of a Master Plan and the implementation of necessary zoning by-law changes identified by that Master Plan.

2.5.8.2 Applicability

This bylaw shall apply to the issuance of all building permits for construction of new dwellings, except for those exemptions and exceptions set forth in subsections 2.5.8.5 and 2.5.8.6.

2.5.8.3 New Dwelling Unit Limitation

Commencing upon the date of adoption of this bylaw, building permits shall not be issued for the construction of, or conversion to, more than thirty-seven (37) new dwelling units in any twelve (12)-month period. Following adoption of this bylaw, the number of units to be allowed in such period shall be determined by subtracting from the thirty-seven (37)-unit limit the total number of permits authorized, minus those permits withdrawn or expired without use.

Building permit applications refused because of this limitation shall be held and acted upon in chronological sequence, based upon the date of receipt by the Building Inspector. In order to ensure a fair distribution of permits, building permits granted to any one (1) applicant in any twelve (12) month period shall be limited to a total of six (6) new dwelling units, unless a special permit is granted pursuant to subsection 2.5.8.6.

2.5.8.4 Procedures and Permit Phasing

No applicant may have more than two (2) complete applications for a new single family building permit before the Building Inspector in any given month. Building permits shall not be issued to any applicant authorizing the construction of more than six (6) new single family dwelling units (exclusive of unused authorizations that have lapsed or have been withdrawn) in any twelve (12)-month period for any lots that are, as of date of adoption of this bylaw, contiguous or held in the same ownership, unless the Planning Board has granted an exemption from this requirement pursuant to a special permit, as provided in subsection 2.5.8.6.

The procedures for issuing the new single family building permits referred to herein shall be as follows:

a. Applications for a building permit must be completed and filed. Improperly filed or rejected permits shall not be deducted from the available units to be authorized in any month, will not be included in the chronological sequence described in 2.5.8.4, and will require a new submittal.

b. The Building Inspector must process and issue permits for all complete and properly filed permit applications pursuant to this bylaw, in the chronological sequence in which such applications are received.

c. To assist the Building Inspector, the Planning Board shall establish procedures, and may adopt reasonable rules and regulations, to ensure the proper administration of this bylaw and the orderly phasing of residential development associated with building permits for any land subdivided pursuant to the provisions of MGL Chapter 41, the Subdivision Control Law.

2.5.8.5 Exemptions

The permit limitation provided in this bylaw shall not apply to building permits for the construction of the following:

a. Any housing unit to be built under any program or statute intended to assist in the construction of low or moderate income or elderly housing as defined in any applicable statute or regulation, including Town bylaws.

b. Reconstruction of an existing dwelling after catastrophic loss, or repairs, expansion, alteration or historic restoration to an existing dwelling.

c. Applications for permits that have been exempted from the limitations of this bylaw, as provided in subsection 2.5.8.6 below.

d. The construction of a single family dwelling on land that, as of the effective date of this bylaw, was part of a lot held in separate ownership and containing one single-family dwelling, provided that only one (1) such new dwelling may be constructed in any year, and provided that no more than two (2) dwellings may be created from the original lot, with the existing and new dwelling each on separate lots, complying with all zoning requirements of the district in which they are located.

e. Permits for non-residential purposes.

2.5.8.6 Special Permit for More Rapid Development

For purposes of this bylaw, the Planning Board shall be the special permit granting authority. Notwithstanding the requirements of subsections 2.5.8.3 and 2.5.8.4 above, the Planning Board may issue a special permit authorizing the immediate issuance of a building permit for up to ten (10) single-family dwelling units upon making all of the following determinations and findings:

a. A salient and unmet housing need would be addressed by the granting of such permit; and

b. Adequate infrastructure and other mitigating measures are being provided to ensure that municipal services will not be overburdened; and

c. Expected benefits to the community outweigh anticipated adverse impacts upon municipal service or public facilities associated with the issuance of a Special Permit.

2.5.8.7 Duration

This by-law shall remain in effect until the earlier of (a) two years after the adoption by the Town of Bolton of a Master Plan or (b) December 31, 2008

2.5.8.8 Zoning Change Protection

The protection against subsequent zoning change provided by MGL c. 40A, §6, to land in a subdivision shall, in the case of a residential development whose completion has been limited by subsection 2.5.8.3 above, be extended by a time period to enable completion.

2.5.8.9 Severability

Each provision of this bylaw shall be construed as separate, to the end that if any part of it shall be held invalid for any reason, the remainder shall continue in full force and effect.

2.5.9 INCLUSIONARY HOUSING

2.5.9.1 Purpose and Intent

The purpose of this Bylaw is to outline and implement a coherent set of policies and objectives for the development of affordable housing in compliance with the Bolton Affordable Housing Plan, MGL c. 40B sec. 20-23 and ongoing programs within the Town of Bolton to promote a reasonable percentage of housing that is affordable to moderate income buyers. It is intended that the Affordable Housing Units (AHU) that result from this Bylaw be considered as Local Initiative Program (LIP) dwelling units in compliance with the requirements for the same as specified by the Department of Community Affairs, Division of Housing and Community Development and that said units count toward the Town's requirements under MGL c. 40B, sec. 20-23.

2.5.9.2 Definitions

1. Affordable Housing Unit (AHU). A dwelling unit available at a cost of no more than 30% of gross household income of households at or below 80% of The Metropolitan Statistical Area (MAS) which includes the Town of Bolton (the Bolton MSA) median income as reported by the U.S. Department of Housing and Urban Development, including units listed under MGL 40B sections 20-24 and the Commonwealth's Local Initiative Program.

2. Qualified affordable housing unit purchaser. An individual or family with household incomes that do not exceed 80% of the median income, with adjustments for household size, as reported by the most recent information from the United States Department of Housing and Urban Development (HUD) and/or the Massachusetts Department of Housing and Community Development (DHCD).

3. Special Permit Granting Authority (SPGA). The SPGA shall either be the Planning Board (if the project is a regular Major Residential Subdivision, FOSPRD or regular development on a combination of ANR and/or Backland lots (and the affordable units considered as Local Initiative Program (LIP) dwelling units) or the Zoning Board of Appeals (ZBA) if the project is brought forth as a "friendly 40B" project and is required to follow Comprehensive Permit guidelines.

4. Off-Site. Off-site shall mean fully buildable lots not contiguous to the primary development as of the date of transfer or application, whichever is applicable, located within the Town of Bolton.

5. Rehabilitated. Pre-existing legal housing stock (no non-conforming pre-existing structures shall be allowed in this definition) located in Bolton that is not under affordable guidelines that has been renovated to current State Building Codes and Bolton Board of Health Rules and Regulations as may be required for the purpose of use as an Affordable Housing Unit.

2.5.9.3 Applicability

1. Division of Land. This Bylaw shall apply to the division of contiguous land held in single or common ownership into eight (8) or more lots. The conditions of this Bylaw will apply to the special permit required and issued for land divisions under MGL c. 40A sec. 9 as well as for "conventional" or "grid" divisions allowed by MGL c. 41 sec. 81-L and sec. 81-U, including those divisions of land that do not require subdivision approval. This bylaw therefore applies to Section 2.3.6 (Farmland and Open Space Planned Residential Development) and Section 2.3.7 (Major Residential Development) of the Zoning Bylaws.

- 2. Multiple Dwelling Units. This Bylaw shall apply to the constructions of eight (8) or more dwelling units, whether on one or more contiguous parcels, and shall require a special permit from the Board of Appeals.
- 3. The provisions of Section 2.5.9.3(2) shall apply to the construction of eight (8) or more dwelling units on individual lots if said eight (8) or more lots held in single or common ownership.
- 4. To prevent the intentional segmentation of projects designated to avoid the requirements of this Bylaw, parcels held in single or common ownership and are subsequently divided into eight (8) or more lots cannot later defeat the requirements of this Bylaw.
- 2.5.9.4 Mandatory Provision of Affordable Units

The SPGA, as a condition of approval of any development referred to in Sections 2.5.9.3, require that the applicant for special permit approval comply with the obligation to provide affordable housing pursuant to this Bylaw and more fully described in Section 2.5.9.5.

2.5.9.5 Provision of Affordable Units

The SPGA shall deny any application for a special permit for development if the Applicant for special permit approval does not comply, at a minimum, with the following requirements for affordable units:

- 1. The maximum housing cost for affordable units created under this Bylaw is as established by the Commonwealth's Division of Housing and Community Development or as revised by the Town.
- 2. One (1) in every eight (8) buildable lots in a division of land or one (1) in every eight (8) units in a multiple unit development subject to this Bylaw shall be established as AHU's in any one or combination of methods provided below. Fractions of a lot or dwelling unit will not be rounded up to the next whole number. For reference, the following schedule is provided for allocating affordable units given a particular range of total lots in a subdivision or total units in a multiple unit development. This schedule is given for reference:

Total Lots / Units	Affordable Lots / Units
8 to 15	1
16 to 23	2
24 to 31	3
32 to 39	4
40 to 47	5
48 to 55	6
56 to 63	7
64 to 71	8
and so o	on

- 3. The AHU(s) shall be constructed or rehabilitated on the locus subject to the special permit.
- 4. The AHU(s) constructed or rehabilitated on a locus different than the one subject to the special permit (see Section 2.5.9.9).

Established

5. an equivalent fees-in-lieu of payment and/or donation of land in fee simple may be made (See Section 2.5.9.12, below).

The applicant may offer, and the SPGA may accept, any combination of the Section 2.5.9.5.(3)-(5) requirements provided that in no event shall the total number of units or land area provided be less than the equivalent number or value of affordable units required by this Bylaw.

2.5.9.6 Provisions Applicable to AHU's On- and Off-Site

1. Siting of AHU's - All affordable units created under this Bylaw shall be situated within the development so as not to be in less desirable locations than market-rate units in the development and shall, on average, be no less accessible to public amenities, such as open space, as the market-rate units.

2. Minimum design and construction standards for affordable units – AHU's within market rate developments shall be integrated with the rest of the development, shall be externally undistinguishable from the market rate units and compatible construction and quality of materials with other units.

3. Timing of construction or provision of affordable units or lots. The SPGA will impose conditions on the special permit requiring construction of affordable housing according to a specified time table, so that, affordable housing units shall be provided coincident to the development of market-rate units, but in no event shall the development of affordable units be delayed beyond the schedule noted below:

MARKET-RATE UNIT (% BUILT) AFFORDABLE HOUSING UNIT (% COMPLETED)

Up to 30%	None required
30% to 50%	At least 30%
50% to 75%	At least 75%
75% or more	100%

Fractional units will be rounded up to the nearest whole number. The actual development schedule for the availability of affordable units and/or the donation of land / fees-in-lieu will be detailed in the special permit decision rendered by the SPGA.

2.5.9.7 Local Preference

The SPGA shall require the applicant to comply with local preference requirements, if any, as established by the Board of Selectmen and/or the Bolton Affordable Housing Partnership.

2.5.9.8 Marketing Plan for Affordable Units

Applicants under this Bylaw shall submit a marketing plan or other method approved by the SPGA, to the SPGA for approval, which describes how the affordable units will be marketed to potential homebuyers. This plan shall include a description of the lottery or other process to be used for selecting buyers. The plan shall be in conformance to DHCD rules and regulations.

2.5.9.9 Provision of Affordable Housing Units Off-Site

As an alternative to the requirements of Section 2.5.9.5(2)-(3), and subject to the approval of the SPGA, an applicant subject to the Bylaw may develop, construct or otherwise provide affordable units equivalent to those required by Section 2.5.9.5(2)-(3) on an alternate site in the Town Of Bolton and approved by SPGA. All requirements of this Bylaw that apply to on-site provision of affordable units, shall apply to provision of off-site affordable units. In addition, the location of the off-site units to be provided shall be approved by the SPGA as an integral element of the special permit review and approval process.

2.5.9.10 Maximum Incomes and Selling Prices: Initial Sale

To ensure that only eligible households purchase AHU's, the purchaser of an affordable unit shall be required to submit copies of the last three years' federal and state income tax returns and certify, in writing and prior to transfer of title, to the developer of the housing units or his/her agent, and within thirty (30) days following transfer of title, to the Bolton Housing Authority or to another authority as determined by the SPGA, that his/her or their family's annual income level does not exceed the maximum level as established by the Commonwealth's Division of Housing and Community Development, and as may be revised from time to time.

The maximum price of the AHU'(s) created under this Bylaw is established by DHCD under the Local Initiative Program (LIP) guidelines in effect at the time the unit(s) is built.

2.5.9.11 Preservation of Affordability; Restrictions on Resale

Each affordable unit created in accordance with this Bylaw shall have the following limitations governing its resale. The purpose of these limitations is to preserve the long-term affordability of the unit and to ensure its continued availability for affordable income households. The resale controls shall be established through a deed restriction, acceptable to DHCD, on the property, recorded at the Worcester County Registry of Deeds or the Land Court, and shall be in force for a period of ninety-nine (99) years.

1. AHU'(s) Resale Price: Sales beyond the initial sale to a qualified purchaser shall not exceed the maximum sales price as determined by the DHCD for affordability within the Town of Bolton at the time of resale.

- 2. Right of first refusal to purchase The purchaser of an AHU developed as a result of this Bylaw shall agree to execute a deed rider prepared by the Town, granting, among other things, the Town of Bolton's right of first refusal for a period not less than one hundred and eighty (180) days to purchase the property or assignment thereof, in the event that, despite diligent efforts to sell the property, a subsequent qualified purchaser cannot be located.
- 3. The SPGA shall require, as a condition for special permit approval under this Bylaw, that the deeds to the AHU's contain a restriction that any subsequent renting or leasing of said AHU shall not exceed the maximum rental price as determined by the DHCD for affordability within the Town of Bolton.
- 4. The SPGA shall require, as a condition for special permit approval under this Bylaw, that the applicant comply with the mandatory set-asides and accompanying deed restrictions on affordability, including the execution of the deed rider noted in this Section 2.5.9.11. The Zoning Enforcement Officer shall not issue a building permit for any affordable unit until the deed restriction is recorded at the Worcester County Registry of Deeds or the Land Court.
- 5. The Bolton Housing Authority or other 501 (C)(3) fund as determined by the SPGA will be the authority that will monitor, oversee and administer the details for all resale of any affordable units created under this Bylaw.
- 2.5.9.12 Donation of Land and/or Fees-in-Lieu of Affordable Housing Unit Provision As an alternative to the requirements of Section 2.5.9.5, and as allowed by law, an applicant may contribute a fee or land to the Bolton Affordable Housing Trust Fund or other 501(C)(3) fund as designated by the SPGA to be used for the development of affordable housing in-lieu of constructing and offering affordable units within the locus of the proposed development or off-site.
 - 1. Calculation of fees-in-lieu of units. The applicant for development subject to this by-law may pay fees in lieu of the construction or provision of affordable units is determined to be \$200,000 per unit. For example, if the applicant is required to construct two affordable income units, he/she may opt to pay \$400,000 in lieu of constructing or providing the units.
 - 2. Schedule of fees-in-lieu of payments. Fees in lieu of payments shall be paid according to the schedule set forth in Section 2.5.9.6(3) above.
 - 3. An applicant may offer, and the SPGA, in concert with the Board of Selectmen, may accept, donations of land in fee simple, on or off-site, that the SPGA determines are suitable for the construction of affordable housing units concurrently or in the future. The value of donated land shall be equal to or greater than 115% of the construction or set-aside of affordable units.

The SPGA shall require, prior to accepting land as satisfaction of the requirements of this by-law, that the applicant submit two (2) appraisals of the land in question (future values may be taken into account in this appraisal), as well as other data relevant to the determination of equivalent value. The applicant must also supply certified information that the land to be donated will support the required number of dwelling units per current Bolton Zoning Bylaws, Bolton Board of Health regulations and Bolton Conservation Commission requirements as of the date of transfer (i.e.; the land is buildable).

2.6 DEFINITIONS:

In this Bylaw the following terms, unless a contrary meaning is required by the context or is specifically prescribed, shall have the following meanings:

Accessory Building or Use - A building or use customarily incidental to and located on the same lot with the principal building or use, except that if more than 50% of the lot area is occupied by such use, it shall no longer be considered accessory.

Assisted Living Housing - Housing units and associated facilities for the elderly, 55 and older, who require daily assistance but who do not require nursing home care. Associated or shared facilities may include common dining facilities, housekeeping services and common space for indoor and outdoor social, educational and recreational activities. Associated facilities may provide additional services beyond Congregate Housing, including daily meals and personal services, medical monitoring and supervision.

Backland Lot - In the Residential Zoning District a lot as permitted by 2.3.5.5 Backland Zoning

Base Map – The Base Town Map and Zoning District Map shall be known as and referred to in the Town of Bolton Zoning Bylaws as the "Base Map". The Base Map is the map prepared by Ducharme & Dillis Civil Design Group, Inc., revised October 31, 2005 and on file with the Town Clerk.

Building - A structure having walls and a roof, including porches covered or uncovered, capable of being used to shelter persons, animals, materials or equipment.

Bulk Storage - Outdoor storage of sand, lumber, coal or other large volume materials, and storage of liquids in tanks except underground as an accessory use, except customary storage in connection with residential use.

Common Driveway - In the Residential Zoning District a driveway serving no more than five single family dwellings or lots as permitted by 2.4.1.2 Common Driveways Serving 3 to 5 Houses.

Congregate Living Housing - Housing units and associated facilities for the elderly, 55 and older, who do not require constant supervision. A Congregate Living Housing unit consists of a room or group of rooms for one or more persons with provisions for living and sleeping for the exclusive use of the individual or household unit. The Congregate Living Housing unit may provide exclusive cooking and sanitary facilities. Associated or shared facilities may include common dining facilities, housekeeping services and common space for indoor and outdoor social, educational and recreational activities.

Dwelling - A building containing one (l) or more rooms with cooking, living, sanitary and sleeping facilities arranged for the use of one (l) or more persons living together as a single housekeeping unit.

Extensive Outdoor Recreation - Premises comprising 10 acres or more used principally for standard golf course, skiing, snowmobiling, horseback riding, camping, or similar extensive recreation, with structures occupying not more than 2% of lot area.

Home Occupation - An occupation or profession engaged in within a dwelling, or accessory building, by a resident thereof as a use accessory thereto.

Lot - a portion of subdivided land, shown on a plan submitted by a Registered Land Surveyor, which conforms to the bylaws of the Town of Bolton.

Lot Area - the horizontal area of the lot exclusive of any area in a street or recorded way open to public use within the Town of Bolton. The minimum lot area for permitted uses within the Town of Bolton must be met by contiguous horizontal area within the Town of Bolton. However, see exception in Note B of Section 2.3.5.2.

Lot Coverage - Percentage of total lot area, within the zone of proposed use, that is covered by structures or roofed.

Lot Frontage - That portion of a lot fronting upon and providing rights of access to a street shown on the Base Map, and listed on the Schedule of Town Roads on file with the Town Clerk, or to an approved or accepted new right-of-way laid out in accordance with the Rules and Regulations Governing the Subdivision of Land in the Town of Bolton, to be measured continuously along a single street line. Owners of lots fronting upon two streets may select that which shall be considered "frontage". On a corner lot frontage shall be measured to the point of intersection of the extension of the side lines of the streets. The minimum frontage for permitted uses within the Town of Bolton must be met by contiguous frontage within the Town of Bolton on an approved or accepted right-of-way in the Town of Bolton.

Mobile Home - A moveable or portable dwelling unit built on a chassis, designed for connection to utilities when in use, and designed without necessity of a permanent foundation for year round living.

Mobile Home Park - Premises which have been planned and improved for the placement of two or more mobile homes for non-transient use.

Municipality - The Town of Bolton.

Nursing Care Facilities - Intermediate and skilled care nursing facilities to provide an intensive level of nurse and medical care for patients.

Parcel - A portion of subdivided land, shown on a plan submitted by a Registered Land Surveyor, which does not conform to the bylaws of the Town of Bolton. The parcel shall be noted as a non-buildable parcel on said plan.

Principal Building or Use - A building or use which is not an accessory use.

Right-of-Way - A public way, or a way which the Town Clerk certifies is maintained and used as a public way, or a way approved and endorsed by the Planning Board in accordance with the Subdivision Control Law, or a way in existence when the Subdivision Control Law became effective in the Town of Bolton.

Shared Driveway - In Residential Zoning District a driveway serving no more than two single-family dwellings or lots as permitted by 2.4.1 Driveways

Sign - Any device displaying, or any display of, any letter, word, picture, symbol, or object designed to inform or attract the attention of persons not on the premises on which such device or display is located, including bill-boards and any such internally or decoratively illuminated building surface other than an unobstructed window.

Structure - A combination of materials assembled by man, the use of which requires fixed location on the ground, or attachment to something on the ground. Structures include buildings, dwellings, towers, masts, sheds, roofed storage areas, mechanical equipment, swimming pools, permanent sporting facilities and signs. Structures do not include fences, walls, driveways, walkways, or other paved areas.

Yard - An area open to the sky, located between a street or other property line and any structure or element thereof other than a fence, wall, other customary yard accessory, or projection allowed to encroach on building lines by the Commonwealth of Massachusetts State Building Code. Depth is to be measured perpendicularly to the street or property line.

Yard, Front - A yard extending between lot side lines across the front of a lot adjacent to each street the lot adjoins.